

CITY COUNCIL

MEETING AGENDA

October 19, 2023

Mark Ammel, Mayor Karen Moore, Mayor Pro Tem Ronald J. Beauchamp, Council Member Tyler DuBord, Council Member Todd Flath, Council Member James R. McNeil, City Manager Phil DeMay, City Clerk Laura J. Genovich, City Attorney

City Council Chambers located at: City Hall - 410 Ludington Street - Room C101 - Escanaba, MI 49829

The Council has adopted a policy to use a Consent Agenda, when appropriate. All items with an asterisk (*) are considered routine by the City Council and will be enacted by one motion. There will be no separate discussion of these items unless a Council Member or citizen so requests, in which event, the item will be removed from the General Order of Business and considered in its normal sequence on the Agenda.

Regular Meeting

Thursday, October 19, 2023, at 7:00 p.m.

CALL TO ORDER

ROLL CALL

INVOCATION/PLEDGE OF ALLEGIANCE

APPROVAL/CORRECTION(S) TO MINUTES – Joint Work Session – October 5, 2023

Regular Meeting – October 5, 2023

APPROVAL/ADJUSTMENTS TO THE AGENDA CONFLICT OF INTEREST DECLARATION(S)
BRIEF PUBLIC COMMENT(S)
PUBLIC HEARINGS – None
UNFINISHED BUSINESS – None
NEW BUSINESS

1. Approval – PFAS Retainer Agreement – Manager.

Explanation: Administration is requesting City Council approval of a retention agreement with Edelson PC and Goodman Acker PC. The firms will represent the city in the ongoing PFAS litigation. Administration sent three proposals to Foster Swift for review, and this agreement was recommended.

2. Approval – Two (2) 2023 GMC Savana Cargo Vans RWD 2500 – Water.

Explanation: Administration is requesting City Council approval to purchase two (2) 2023 GMC Savana Cargo Vans RWD 2500 complete with ladder racks and shelving from Todd Wenzel Buick-GMC of Westland, Michigan at a cost of \$82,764.00 (\$41,382.00 per cargo van) with contingencies of \$5,236 (\$2,618.00 per cargo van) to include signage, amber lighting, and city 2-way radios in an amount not to exceed \$88,000.00. Money is not budgeted, but is available for this purchase in the water fund.

3. Approval – Engineering Services for DWSRF – ARPA-2 Project – Water.

Explanation: Administration is requesting City Council approval to retain C2AE from Escanaba, MI, for engineering services with regard to the City of Escanaba DWSRF ARPA-2 Project in an amount not to exceed \$2,320,900.00.

4. Approval – Engineering Services for WTP Filter Bed Rehabilitation – Water.

Explanation: Administration is requesting City Council approval to retain C2AE from Escanaba, MI, for engineering services with regard to the City of Escanaba Water Treatment Plant Filter Bed Rehabilitation in an amount not to exceed \$8,000.00. Money is not budgeted, but is available in the water fund under professional services for this type of work.

5. Approval – Ludington Lift Station Inspection - Asbestos – Wastewater.

Explanation: Administration is requesting City Council approval to obtain the services of Pearson Asbestos Abatement of Escanaba, Michigan to carry out non-destructive inspection of the Ludington Lift Station building for asbestos containing materials and paint for lead and cadmium in an amount not to exceed \$2,000.00. Money is budgeted and available for this type of activity.

6. Approval – Resolution 23-22 – Termination of Participation in MERS Defined Benefit Plan - Manager.

Explanation: Administration is requesting City Council approval of a resolution terminating participation in the MERS Defined Benefit Plan. This resolution is in accordance with the MERS Plan Termination Policy. Assets, liabilities, and fiduciary duties will transfer to the successor plan.

7. Approval – Resolution 23-23 – Adoption of City of Escanaba Defined Benefit Retirement Plan - Manager.

Explanation: Administration is requesting City Council approval of a resolution adopting all the components of the City of Escanaba Defined Benefit Retirement Plan. This approval includes the Plan document, establishment of the Retirement Board, Board Rules of Procedure, and Statement of Investment Policy.

APPOINTMENTS
BOARD, COMMISSION, AND COMMITTEE REPORTS
GENERAL PUBLIC COMMENT
ANNOUNCEMENTS
ADJOURNMENT

Respectfully Submitted,

Yames Welliel

James R. McNeil City Manager

OFFICIAL PROCEEDINGS CITY COUNCIL WELLS TOWNSHIP CITY OF ESCANABA, MICHIGAN Joint Work Session Meeting Thursday, October 5, 2023

Pursuit to a special meeting notice posted September 28, 2023, the meeting was called to order by the Honorable Mayor Mark Ammel at 5:00 p.m.

Present: Mayor Mark Ammel, Council Members, Ronald J. Beauchamp, Tyler

DuBord, and Karen Moore

Absent: Council Member Todd Flath

Present: Supervisor Bob Stasewich, Treasurer Karen Kobasic, Trustees, Brett Way,

Gloria F. Johnson, and Dorothy J. Pepin

Absent: Clerk Susan Mroczkwski and Trustee Judith Gereau

Also Present: City Manager James R. McNeil, City Clerk Phil DeMay, Department

Heads, media, and members of the public.

DuBord moved, Beauchamp seconded, **CARRIED UNANIMOUSLY**, to excuse Council Member Flath and Wells Twp Clerk Susan Mroczkwski and Trustee Judith Gereau.

ADJUSTMENTS TO THE AGENDA

DuBord moved, Beauchamp seconded, **CARRIED UNANIMOUSLY**, to approve the City Council Agenda as submitted.

CONFLICT OF INTEREST DECLARATION – None

NEW BUSINESS

NB-1 Discussion – City of Escanaba – Wells Township Sanitary Sewer Expansion.

City and Wells Township administration led a discussion on a proposal to expand sanitary sewer into the township. Time was allotted for a presentation, question and answer, and public comment.

GENERAL PUBLIC COMMENT – None

ANNOUNCEMENTS – None

Hearing no further public comment, Council adjourned at 5:54 p.m.	, DuBord	moved,	Trustee	Way	seconded,	the
Respectfully submitted,						
Phil DeMay	Approved	:				
City Clerk			k Ammel,	May	or	

OFFICIAL PROCEEDINGS CITY COUNCIL CITY OF ESCANABA, MICHIGAN Regular Council Meeting Thursday, October 5, 2023

The meeting was called to order by the Honorable Mayor Mark Ammel at 7:00 p.m. in the Council Chambers of City Hall located at 410 Ludington Street.

Present: Mayor Mark Ammel, Council Members, Ronald J. Beauchamp, Tyler

DuBord, and Karen Moore

Absent: Council Member Todd Flath

Also Present: City Manager James R. McNeil, City Clerk Phil DeMay, Department Heads,

media, and members of the public.

City Clerk DeMay led Council in the Pledge of Allegiance.

DuBord moved, Beauchamp seconded, **CARRIED UNANIMOUSLY**, to excuse Council Member Flath's absence.

Moore moved, DuBord seconded, **CARRIED UNANIMOUSLY**, to approve Regular Meeting minutes from September 21, 2023, as submitted.

ADJUSTMENTS TO THE AGENDA

Beauchamp moved, DuBord seconded, **CARRIED UNANIMOUSLY**, to approve the City Council Agenda as submitted.

CONFLICT OF INTEREST DECLARATION – None

BRIEF PUBLIC COMMENT

- Craig Woerpel Discussed Spooktacular Trunk or Treat Sunday October 22nd 4-6pm.
- Ed Legault Discussed City Manager review.

PUBLIC HEARINGS

<u>PH-1 Public Hearing - Municipal Employees' Retirement System (MERS) Termination - City Manager.</u>

In accordance with the MERS Plan Termination Policy, this was the first of two consecutive public meetings to discuss termination of the MERS Defined Benefit Pension Plan. At the next council meeting, administration will request adoption of a termination resolution and approval of the successor plan named "City of Escanaba"

Defined Benefit Retirement Plan."

City Administration and City Council gave an opportunity for MERS representatives to speak on what they offer to the employee retirement system. Along with answering questions or concerns.

City Manager James McNeil presented to Council a review of why administration is suggesting a termination from MERS.

This being a public hearing, Mayor Ammel asked for public comment.

Hearing no public comment, Mayor Ammel then closed the public hearing.

UNFINISHED BUSINESS – None

NEW BUSINESS

NB-1 Approval - Fire Hydrant Painting - Water Department.

Administration requested City Council approval to hire Markley's Precision Company LLC to paint fire hydrants at \$515 per hydrant with lead abatement and \$410 per hydrant with non-lead in an amount not to exceed \$80,000. Money was available and budgeted for this activity.

NB-1 DuBord moved, Beauchamp seconded, to approve to hire Markley's Percision Company LLC to paint fire hydrants at \$515 per hydrant with lead abatement and \$410 per hydrant with non-lead in an amount not to exceed \$80,000.

Upon a call of the roll, the vote was as follows:

Aves: DuBord, Beauchamp, Moore, Mayor Ammel

Nays: None Absent: Flath

MOTION CARRIED.

NB-2 Approval - Engineering Inspection of Antennas of South Water Tower - Water Department.

Administration requested City Council approval to retain the services of Dixon Engineering and Inspection Services to perform an inspection of NMU antennas of South Water Tower in an amount not to exceed \$1,450. NMU will fully reimburse the City for these costs.

NB-2 Moore moved, DuBord seconded, to approve to retain the services of Dixon Engineering and Inspection Services to perform an inspection of NMU antennas of South Water Tower in an amount not to exceed \$1,450.

Upon a call of the roll, the vote was as follows:

Ayes: Moore, DuBord, Beauchamp, Mayor Ammel

Nays: None Absent: Flath

MOTION CARRIED.

NB-3 Approval – WTP Asphalt Patching – Water Department.

Administration requested City Council approval to obtain the services of Payne and Dolan of Gladstone, MI to complete the needed asphalt patchwork created from work done on the water system in an amount not to exceed \$590.00/ton. Money was budgeted and available.

NB-3 Moore moved, DuBord seconded, to approve to obtain the services of Payne and Dolan of Gladstone, MI to complete the needed asphalt patchwork created from work done on the water system in an amount not to exceed \$590.00/ton.

Upon a call of the roll, the vote was as follows:

Ayes: Moore, DuBord, Beauchamp, Mayor Ammel

Nays: None Absent: Flath

MOTION CARRIED.

NB-4 Approval – Use of Public Space – Spooktacular Trunk or Treat - DDA

Escanaba DDA sought City Council approval to close South 11th Street from Ludington Street to 1st Avenue South and North 11th Street from Ludington Street to 1st Avenue North for a Spooktacular Trunk or Treat event on Sunday, October 22, 2023, from 3:00 p.m. to 6:30 p.m. (Event Time: 4:00 p.m. to 6:00 p.m.) Administration recommended approval with the following conditions: 1) Proper insurance be provided naming the City of Escanaba as an additional insured, 2) The event sponsors provide all labor and material to clean up at the conclusion of the event.

NB-4 Moore moved, Beauchamp seconded, **CARRIED UNANIMOUSLY**, to approve a request from the Escanaba DDA to close South 11th Street from Ludington Street to 1st Avenue South and North 11th Street from Ludington Street to 1st Avenue North, along with Ludington intersection for a Spooktacular Trunk or Treat event on Sunday, October 22, 2023, from 3:00 p.m. to 6:30 p.m. (Event Time: 4:00 p.m. to 6:00 p.m.), provided: 1) Proper insurance be provided naming the City of Escanaba as an additional insured, 2) The event sponsors provide all labor and material to clean up at the conclusion of the event.

NB-5 Presentation – Summary Report – Ogden Triangle Historic Resource Inventory – Planning & Zoning.

At their regular meeting on September 18, 2023, the Historic District Commission moved to forward the Summary Report for the Ogden Triangle Historic Resource Inventory to City Council for their review and comments.

NB-6 Discussion - City Manager Review - City Council.

City Council discussed the City Manager's review as outlined in the contract.

City Council members felt that it was not necessary to go through an official written review. Council members gave rave reviews on the City Managers performance and commended him on his ability to achieve above and beyond his set goals in a year.

Moore moved, Beauchamp seconded, **CARRIED UNANIMOUSLY**, to give an official 4 performance review to be recorded in his personnel file.

Absent: Flath

APPOINTMENT(S) TO CITY BOARDS, COMMISSIONS, AND COMMITTEES

Manager McNeil, with Council Consensus made the following appointment:

Amy Reddinger appointed to the Board of Library Trustees, expiring June 1, 2025.

BOARD, COMMISSION, AND COMMITTEE REPORTS

Council Members reviewed City Board and Commission meetings each attended since the last City Council Meeting.

GENERAL PUBLIC COMMENT

- Craig Woerpel discussed grant funding to revitalize downtown Escanaba.
- Tyler Anthony discussed appointments for ZBA.

ANNOUNCEMENTS

- Stephenson Ave. is paved! Great job by all the workers completing the project!
- Council Election is coming up November 7, 2023!

October 5, 2023 – cont.	
Hearing no further public commen p.m.	, DuBord moved, the Council adjourned at 8:32
Respectfully submitted,	
Phil DeMay City Clerk	Approved: Mark Ammel, Mayor

City Council Minutes

Agenda Item: 📐 Date: 10-10-

City Council Agenda Item Request

Date: 10/10/2023

Name: James McNeil

Department: Manager

Item: PFAS Retainer Agreement

Meeting date requested: 10/19/2023

Explanation for request: Administration is requesting City Council approval of a retention agreement with Edelson PC and Goodman Acker PC. The firms will represent the city in the ongoing PFAS litigation. Administration sent three proposals to Foster Swift for for review, and this agreement was recommended.

Agenda Item: <u>NB-2</u>
Date: <u>10-19-2023</u>

City Council Agenda Item Request

Date: 10/12/2023

Name: Jeff Lampi

Department: Water Department

Item: Two (2) 2023 GMC Savana Cargo Vans RWD 2500

Meeting date requested: 10/19/2023

Explanation for request:

Admnistration is requesting City Councils approval to purchase two (2) 2023 GMC Savana Cargo Vans RWD 2500 complete with ladder racks and shelving from Todd Wenzel Buick-GMC of Westland, Michigan at a cost of \$82,764.00 (\$41,382.00 per cargo van) with contingencies of \$5,236 (\$2,618.00 per cargo van) to include signage, amber lighting, and city 2-way radios in an mount not to exceed \$88,000.00.

Money is not budgeted, but is available for this purchase in the water fund.

City of Escanaba

Prepared For: Jeff Lampi

jlampi@escanaba.org

Vehicle: [Fleet] 2023 GMC Savana Cargo Van (TG23405) RWD 2500 135"

TODD WENZEL BUICK GMC of WESTLAND

35100 FORD RD WESTLAND, MI 48185

State of Michigan MiDeal Contract # 071B7700184 MiDeal Spec # 0124-VAN

City of Escanaba Pricing (MiDeal Pricing)

2023 GMC Savana 2500 Std Cargo Van

8600# GVWR 4.3L V6 Gas / 8-Spd Auto Trans Std Length

MSRP \$ 41,200.00

2023 GMC Savana 2500 Standard Cargo Van		\$ 3	34,837.00
MBP – Masterack Bulkhead Partition (Steel)		\$	895.00
MSP – Masterack Commercial Shelving Package		\$	1,865.00
LCL – LED Cargo Area Light		\$	385.00
DEL – Delivery per MiDeal (Mackinaw City, MI)		\$	515.00
,	Total	\$ 3	88,497.00

Options:

LR0 - KargoMaster Double Clamp Down Ladder Rack + \$ 1,475.00

AU1382

LR1 – Prime Design Single Drop-Down Ladder Rack + \$ 2,475.00

LR2 – Prime Design Dbl Drop-Down Ladder Rack +\$2,885.00

Prices Quoted are for a GM MiDeal Dealer Stock Unit. Available for Immediate Upfit; Availability Subject to Prior Sale. FOB Mackinaw City, MI Vehicle: [Fleet] 2023 GMC Savana Cargo Van (TG23405) RWD 2500 135" (Complete)

Price Summary

RICE SUMMARY		MSRP
Base Price		\$37,500.00
Total Options		\$2,005.00
Vehicle Subtotal		\$39,505.00
Destination Charge	- common contaminant agreement agreement of the first of contaminant and contaminant and contaminant agreement agree	\$1,695.00
Grand Total		\$41,200.00

Oct 11, 2023 Page 2

At the user's request, prices for this vehicle have been formulated on the basis of Initial Pricing for the vehicle, however GM cannot guarantee that Initial Pricing is available. This document contains information considered Confidential between GM and its Clients uniquely. The Information provided is not intended for public disclosure. Prices, specifications, and availability are subject to change without notice, and do not include certain fees, taxes and charges that may be required by taw or vary by manufacturer or region. Performance figures are guidelines only, and actual performance may vary. Photos may not represent actual vehicles or exact configurations. Content based on report preparer's input is subject to the accuracy of the input provided.

Data Version: 20672. Data Updated: Oct 10, 2023 6:44:00 PM PDT.



Albert Li | (734) 713-1065 | ali@toddwenzel.com

Vehicle: [Fleet] 2023 GMC Savana Cargo Van (TG23405) RWD 2500 135" (◆ Complete)

Selected Model and Options

-65				
			_	
ma	£ 31	ാല	-	я.
201	O	-	_	

CODE

MODEL

TG23405

2023 GMC Savana Cargo Van RWD 2500 135"

COLORS

CODE

DESCRIPTION

GAZ

Summit White

GAZ	Summit white
OPTIONS	
CODE	DESCRIPTION
ZW9	Body, standard
FE9	Emissions, Federal requirements
LV1	Engine, 4.3L V6
M5U	Transmission, 8-speed automatic, electronically controlled
GU6	Rear axle, 3.42 ratio
E24	Door, swing-out passenger-side, 60/40 split
1WT	Work Van Preferred Equipment Group
ZLP	Tire, spare LT245/75R16 all-season, blackwall
ZY1	Paint, solid
GAZ	Summit White
AS5	Seats, front bucket with Custom Cloth trim
93G	Medium Pewter, Custom Cloth Seat Trim
ZX2	Seating arrangement, driver and front passenger high-back buckets,
UOH	Audio system, AM/FM stereo with MP3 player and USB port
ZQ3	Driver Convenience Package
G80	Differential, heavy-duty locking rear
UA1	Battery, heavy-duty 770 cold-cranking amps, maintenance-free
KG4	Alternator, 150 amps
Z82	Trailering equipment, heavy-duty
UY7	Trailering provisions, trailering wire harness only

At the user's request, prices for this vehicle have been formulated on the basis of Initial Pricing for the vehicle, however GM cannot guarantee that Initial Pricing is available. This document contains information considered Confidential between GM and its Clients uniquely. The information provided is not intended for public disclosure. Prices, specifications, and availability are subject to change without notice, and do not include certain fees, taxes and charges that may be required by law or vary by manufacturer or region. Performance figures are guidelines only, and actual performance may vary. Photos may not represent actual vehicles or exact configurations. Content based on report preparer's input is subject to the accuracy of the input provided.

Data Version: 20672. Data Updated: Oct 10, 2023 6:44:00 PM PDT.



Albert Li | (734) 713-1065 | ali@toddwenzel.com

Vehicle: [Fleet] 2023 GMC Savana Cargo Van (TG23405) RWD 2500 135" (& Complete)

OPTIONS	
CODE	DESCRIPTION
A12	Glass, fixed rear doors windows
A13	Glass, fixed rear side door windows
ZW3	Glass, rear doors and side cargo doors
ASB	Window security bar, rear side door glass
U2K	SiriusXM Radio
UPF	Bluetooth for phone, personal cell phone connectivity to vehicle audio system
BA3	Console, engine cover with swing-out storage bin
NP5	Steering wheel, leather-wrapped
W1Y	Steering wheel controls, mounted audio controls
ATG	Remote Keyless Entry
K34	Cruise control
UTJ	Theft Alarm Notification
USR	Audio system feature, USB port
	Visors, driver and front passenger, cloth, padded
8S3	Backup alarm, 97 decibels

Options Total

At the user's request, prices for this vehicle have been formulated on the basis of Initial Pricing for the vehicle, however GM cannot guarantee that Initial Pricing is available. This document contains information considered Confidential between GM and its Clients uniquely. The information provided is not intended for public disclosure. Prices, specifications, and availability are subject to change without notice, and do not include certain fees, taxes and charges that may be required by law or vary by manufacturer or region. Performance figures are guidelines only, and actual performance may vary. Photos may not represent actual vehicles or exact configurations. Content based on report preparer's input is subject to the accuracy of the input provided.

Data Version: 20672. Data Updated: Oct 10, 2023 6:44:00 PM PDT.

Oct 11, 2023 Page 4

Vehicle: [Fleet] 2023 GMC Savana Cargo Van (TG23405) RWD 2500 135" (Complete)

Standard Equipment

Standard Eq	
Package	
	Power Convenience Package includes (A31) power windows and (AU3) power door locks
Mechanical	
A DESIGNATION OF THE SECOND	Engine, 4.3L V6 with Direct Injection and Variable Valve Timing, includes aluminum block construction (276 hp [206 kW] @ 5200 rpm, 298 lb-ft of torque [404 Nm] @ 3900 rpm) (STD) (Reference the Engine/Axle page for availability.)
	Transmission, 8-speed automatic, electronically controlled with overdrive and tow/haul mode. Includes Cruise Grade Braking, Powertrain Grade Braking and Tap-Up/Tap-Down Driver Shift Control (STD) (Requires (LV1) 4.3L gas V6 engine. Reference the Engine/Axle page for availability.)
Maga Basis.	Body, standard (STD)
ц « ц — удра марын ролды « компониция, ко уния болбанскогорого послубуй». На найм иден уния болого	GVWR, 8600 lbs. (3901 kg) (Reference the Engine/Axle page for availability.)
	Rear axle, 3.42 ratio (Reference the Engine/Axle page for availability.)
Salam (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971) (1971)	Rear wheel drive
and the second control of the second control of the	Cooling, external engine oil cooler
The state of the s	Battery, 600 cold-cranking amps, maintenance-free with rundown protection and retained accessory power
April in regulation and account of publicating region	Alternator, 105 amps (Not available with (Y3H) Paratransit Package, (C69) rear air conditioning or (9L7) Wiring junction block equipment accessory.)
	Frame, full-length box ladder-type
	Suspension, front independent with coil springs and stabilizer bar
	Suspension, rear hypoid drive axle with multi-leaf springs
	Steering, power
	Brakes, 4-wheel antilock, 4-wheel disc
CALL MARKET CONTRACTOR OF THE	Fuel tank capacity, mid-frame and approximately 31 gallons (117.3L)
	Exhaust, aluminized stainless-steel muffler and tailpipe
Exterior	
	Door, swing-out passenger-side, 60/40 split (STD)
	Wheels, 4 - 16" x 6.5" (40.6 cm x 16.5 cm) steel includes Gray center caps and steel spare
and and the state of the state	Tire, spare LT245/75R16 all-season, blackwall located at rear underbody of vehicle (STD)
Abbits - Asid Art (Art July John) - Angle Art (Asid Art	Tires, front LT245/75R16E all-season, blackwall
те на установа на применения на применения на применения на применения на применения на применения на применен	Tires, rear LT245/75R16E all-season, blackwall
	Bumpers, front and rear painted Black with step-pad

At the user's request, prices for this vehicle have been formulated on the basis of Initial Pricing for the vehicle, however GM cannot guarantee that Initial Pricing is available. This document contains information considered Confidential between GM and its Clients uniquely. The information provided is not intended for public disclosure. Prices, specifications, and availability are subject to change without notice, and do not include certain fees, taxes and charges that may be required by law or vary by manufacturer or region. Performance figures are guidelines only, and actual performance may vary. Photos may not represent actual vehicles or exact configurations. Content based on report preparer's input is subject to the accuracy of the input provided.

Data Version: 20672. Data Updated: Oct 10, 2023 6:44:00 PM PDT.

Page 5

Oct 11, 2023



Albert Li | (734) 713-1065 | ali@toddwenzel.com

Vehicle: [Fleet] 2023 GMC Savana Cargo Van (TG23405) RWD 2500 135" (✔ Complete)

Exterior	
	Headlamps, dual halogen composite
the modern of a first to the fi	Mirrors, outside heated power-adjustable, Black, manual-folding
ne neg kris versicht der eine dem ist Ermenselse und Abstättig Abstät ist Erzel beschändigen mit verbre der mei	Glass, Solar-Ray light-tinted, all windows
and a street and administration of the control of t	Wipers, front intermittent wet-arm with pulse washers
ATENERAL TO COMMITTE CONTRACTOR C	License plate kit, front
Entertainment 🐪	
గు కార్లు చేస్తారు. మార్చికే కట్టుముందు -	Audio system, AM/FM stereo with MP3 player seek-and-scan, digital clock, TheftLock, random select, auxiliary jack and 2 front door speakers (STD) (Not available with (U2K) SiriusXM Radio.)
Vin val. val.	Wi-Fi Hotspot capable (Included and only available with (UE1) OnStar. Terms and limitations apply. See onstar.com or dealer for details.)
Interior	
	Seats, front bucket with vinyl trim and head restraints, includes inboard armrests (STD) (Requires (**W) interior trim.)
Messen gage gewingsgegige in man een peer	Seating arrangement, driver and front passenger high-back buckets, with head restraints and vinyl or cloth trim (STD) (Not available with (AJ3) driver-side only frontal airbag.)
	Console, engine cover with open storage bin
	Floor covering, full-length Black rubberized-vinyl (Not available with (RFM) rear floor covering delete or (PCH) Hotel Shuttle Package.)
	Steering wheel, steel sleeve column with theft-deterrent locking feature, Black
y a room, amin'ny faritr'i Paris ao amin'ny faritr'i Ara-da-dra dia mandritry ao amin'ny faritr'i Ara-da-da-da-da-da-da-da-da-da-da-da-da-da	Instrumentation, analog with speedometer, odometer with trip odometer, fuel level, voltmeter, engine temperature and oil pressure
gelder der verstere der verster der der der der der der der der der d	Oil life monitor
	Warning tones, headlamp on and key-in-ignition
	Driver Information Center includes fuel range, average speed, oil life, tire pressure monitoring, fuel used, ice warning, engine hours, average fuel economy, tachometer, and maintenance reminders. Compass and outside temperature available if equipped.
The side of a last of the Administration of the side o	Door locks, power with lock-out protection (Included with (ZQ2) Power Convenience Package. NOTE: Does no include (ATG) Remote Keyless Entry. (ATG) Remote Keyless Entry must be ordered separately.)
is provided to commonweal Standardstand, quicks — baded — design — 1.50	Windows, power (Included with (ZQ2) Convenience Package.)
	Tow/haul mode selector, instrument panel-mounted
de de la company	Theft-deterrent system, vehicle, PASS-Key III
n englische Antonio en entfact BATELLE Charles en C.C.	Power outlets, 2 auxiliary on engine console cover with covers, 12-volt
1865 kirgit 1866 ki ki kiraman ngan nastanga minaganas at at ada da bibana yang salah Malama	Power outlet, 120-volt

At the user's request, prices for this vehicle have been formulated on the basis of Initial Pricing for the vehicle, however GM cannot guarantee that Initial Pricing is available. This document contains information considered Confidential between GM and its Clients uniquely. The information provided is not intended for public disclosure. Prices, specifications, and availability are subject to change without notice, and do not include certain fees, taxes and charges that may be required by law or vary by manufacturer or region. Performance figures are guidelines only, and actual performance may vary. Photos may not represent actual vehicles or exact configurations. Content based on report preparer's input is subject to the accuracy of the input provided.

Data Version: 20672. Data Updated: Oct 10, 2023 6:44:00 PM PDT.

Oct 11, 2023 Page 6



Albert Li | (734) 713-1065 | ali@toddwenzel.com

Interior	
	Air conditioning, single-zone manual
udus dan 146. 1966 - ₁₉₆ 6. 1765. 17. Bilder och der 1776. 17. Produkt in Bernette bestättlich der kantoniere in	Defogger, side windows
and the second seco	Cup holders, 3 on the engine console cover
MATERIA (1994) I III (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994) (1994)	Trim panels, hinged rear side doors and hinged rear doors (Not available with (Y3H) Paratransit Package.)
er in grif Virologgiesgeber seit i Deur trock til bestätte virologie. All trocket i trocket pieter sisse virologie	Mirror, inside rearview includes Rear Vision Camera display
47 M Po s a latter man ext. o annual control and annual control and annual specific and a spec	Headliner, cloth, over driver and passenger
and the second of the second production of the second of t	Visors, driver and front passenger, vinyl
1994 - Adulation of the proper property and appropriate and property of the pr	Assist handles, driver and right-front passenger
	Lighting, interior with 4 dome lights 1 overhead and 3 LED's on cross members in cargo area, includes defeat switch and door handle-activated switches
and the state of t	Cargo tie-downs 6 D-ring attachments on cargo area floor (Not available with (Y3H) Paratransit Package or (ZP6) 5-passenger Savana Crew Van.)
landalanda a la perioria (per la promissa del primer a la constantina e experimenta e este e en el constantin	GMC Connected Access capable (Included and only available with (UE1) OnStar. Subject to terms. See onstar.com or dealer for details.)
Safety-Mechanical	
ruit disposation and a commission of the	Brake/transmission shift interlock, for automatic transmissions
the property to the second of the comment of the second or second	Hill Start Assist
	StabiliTrak, traction assistance and vehicle stability enhancement system
Safety-Exterior	
n year ya Tibogogo ya di Lifebina ya jinee hay ta ya ba ka Milianda ka	Door beams, steel-side
in the second of	Daytime Running Lamps
Safety-Interior	
	Airbags, frontal, front driver and right front passenger includes passenger-side frontal airbag deactivation switch. (Included with (ZX2) driver and right-front passenger high-back bucket seats or (ZP6) 5-passenger Savana Crew Van Package. Not available with (ZX1) driver only high-back bucket seat or (AJ3) driver-side only frontal airbag. Always use seat belts and child restraints. Children are safer when properly secured in a rear seat in the appropriate child restraint. See the Owner's Manual for more information.)
COMMITTED TO STATE OF THE STATE	Airbags, seat-mounted side-impact for driver and right-front passenger and roof-rail mounted head-curtain for 1-

Airbags, seat-mounted side-impact for driver and right-front passenger and roof-rail mounted head-curtain for 1row coverage with (ZX1) driver only high-back bucket seat or (ZX2) driver and right-front passenger high-back
bucket seats, and 2-row coverage with (ZP6) 5-passenger Savana Crew Van Package (Always use seat belts
and child restraints. Children are safer when properly secured in a rear seat in the appropriate child restraint.
See the Owner's Manual for more information.)

This document contains information considered Confidential between GM and its Clients uniquely. The information provided is not intended for public disclosure. Prices, specifications, and availability are subject to change without notice, and do not include certain fees, taxes and charges that may be required by law or vary by manufacturer or region. Performance figures are guidelines only, and actual performance may vary. Photos may not represent actual vehicles or exact configurations. Content based on report preparer's input is subject to the accuracy of the input provided.

Data Version: 20672. Data Updated: Oct 10, 2023 6:44:00 PM PDT.

Oct 11, 2023 Page 7



Albert Li | (734) 713-1065 | ali@toddwenzel.com

Vehicle: [Fleet] 2023 GMC Savana Cargo Van (TG23405) RWD 2500 135" (Complete)

Safety-Interior

Airbag deactivation switch, frontal passenger-side (Not available with (AJ3) driver only airbag. Always use seat belts and child restraints. Children are safer when properly secured in a rear seat in the appropriate child restraint. See the Owner's Manual for more information.)

OnStar and GMC connected services capable (Terms and limitations apply. See onstar.com or dealer for details.)

Rear Vision Camera display integrated into rearview mirror (Display included with (DRJ) inside rearview mirror.)

Tire Pressure Monitoring System (does not apply to spare tire)

WARRANTY

Basic Years: 3

Basic Miles/km: 36,000 Drivetrain Years: 5 Drivetrain Miles/km: 60,000

Drivetrain Natas Cartain assuranced accomment and qualified floats:

Drivetrain Note: Certain commercial, government, and qualified fleet vehicles: 5 years/100,000 miles

Corrosion Years (Rust-Through): 6

Corrosion Years: 3

Corrosion Miles/km (Rust-Through): 100,000

Corrosion Miles/km: 36,000 Roadside Assistance Years: 5

Roadside Assistance Miles/km: 60,000

Roadside Assistance Note: Certain commercial, government, and qualified fleet vehicles: 5 years/100,000 miles

Maintenance Note: First Visit: 12 Months/12,000 Miles

At the user's request, prices for this vehicle have been formulated on the basis of Initial Pricing for the vehicle, however GM cannot guarantee that Initial Pricing is available. This document contains information considered Confidential between GM and its Clients uniquely. The information provided is not intended for public disclosure. Prices, specifications, and availability are subject to change without notice, and do not include certain fees, taxes and charges that may be required by law or vary by manufacturer or region. Performance figures are guidelines only, and actual performance may vary. Photos may not represent actual vehicles or exact configurations. Content based on report preparer's input is subject to the accuracy of the input provided.

Data Version: 20672. Data Updated: Oct 10, 2023 6:44:00 PM PDT.

Oct 11, 2023

Page 8

ErgoRack

PRIME DESIGN

For GM Vans







Easy to use

Deploys with only 2 fingers.



Improve productivity

Deploy or stow in less than 20 seconds.



Enhance Safety

Eliminates slip and fall hazards from climbing on bumper or tire.



No Drill Installation

Maintain vehicle warranty and resale value with no holes to drill.



ErgoRack

PRIME DESIGN

A Safe Fleet Brand

For Gm Vans







VCC-GM
Dual Clamp

VRR-GM
Dual Rotation

VCR3-GM

Clamp / Rotation with 3rd crossbar

Models	Vehicle Wheelbase	Vehicle Height	Street side Feature	Curb side Feature	Weight
VBB-GM	135 IN	84 IN	End Stop	End Stop	25 lbs
VBC-GM VCB-GM	135 IN	84 IN	End Stop	Clamp	40 lbs
VBR-GM VRB-GM	135 IN	84 IN	End Stop	Rotation	67 lbs
VCC-GM	135 IN	84 IN	Clamp	Clamp	55 lbs
VCR-GM VRC-GM	135 IN	84 IN	Clamp	Rotation	82 lbs
VRR-GM	135 IN	84 IN	Rotation	Rotation	110 lbs
VCC3-GM*	135 IN	84 IN	Clamp	Clamp	65 lbs
VCR3-GM* VRC3-GM*	135 IN	84 IN	Clamp	Rotation	92 lbs
VRR3-GM*	135 IN	84 IN	Rotation	Rotation	120 lbs

^{*}Three crossbar rack

Notes:

Crossbar spacing and the vehicle's wheelbase length will dictate the ladders that can be carried. Crossbars can be spaced at five, six, seven, or eight feet.

A 3rd crossbar can be spaced four, five, six, or seven feet.

If carrying a long extension ladder and a short step ladder, a three crossbar rack is suggested.

Contact Prime Design customer service for application specific questions.

 $Check the \ vehicle \ manufacturer's \ recommendation for \ roof \ capacity \ weight \ limits.$

A wide range of accessories are available including conduit carriers, light holders, and more.

Van Crossbar Spacing and Ladder Size Guide

Step Ladder Size	Extension Ladder Size	
6 – 8 ft.	16-20 ft.	
8 – 10 ft.	16 – 24 ft.	
8 – 12 ft.†	16 – 28 ft.	
10 – 12 ft.†	20 – 32 ft.	
10 – 12 ft.†	20-36 ft.	
	6-8 ft. 8-10 ft. 8-12 ft.† 10-12 ft.'	

^{†12} ft. step ladders may require an extended upper add-on component.





Copyright ©2020 Safe Fleet and its subsidiaries. All rights reserved. No part of this publication may be reproduced by any means without written permission from Safe Fleet. The information in this publication is believed to be accurate. However, Safe Fleet does not make any representation or warranty to that effect and does not assume responsibility for any consequences resulting from use of such information. Revisions or new editions of the publication may be issued (or not issued) in our discretion to incorporate such changes.









CARGO VAN BASE SHELVING PACKAGE - FORD TRANSIT LOW ROOF, GM EXPRESS/SAVANA

Part #02R104KP

If you don't know where to start with your cargo van upfit, the Masterack base shelving package is a great place to start. Comes with three high-quality, steel-constructed tapered shelving modules you can customize with dividers, hooks, drawers, or other accessories specific to your cargo organization needs.

FOR FLEET & DISTRIBUTORS

ADD TO QUOTE

FOR VEHICLE OWNERS

Products Build Quote Track Order Menu

View Specifications

• Vehicle: Ford Transit Low Roof, GM Express/Savana

• Year: N/A

• Trade: Base

Wheelbase: N/A

Material: Steel

• Dimensions: N/A

• Weight: 194

• Install Time: 3 hours, 18 minutes

View Resources

Included in this Package

Products Build Quote Track Order



We've Got You Covered.

Masterack's knowledgeable sales and client services team is available to answer your questions and help with any product-related questions. We're just a phone call or email away

LET'S TALK

About Us

Products Build Quote Track Order

repair, pest control and insurance carriers across the U.S. and continues to do so today.

Contact Us

350 Technology Parkway Suite 400 Peachtree Corners, GA 30092

& 800-222-8785

eNewsletter

your@email.com

SIGN UP





Privacy Policy

California Consumer Privacy Act

Terms & Conditions

Website Accessibility

© 2023 Masterack, LLC. All Rights Reserved

A Business Unit of JB POINDEXTER & CO

Products Build Quote Track Order

Agenda Item: <u>NR-3</u>
Date: <u>10-19-3023</u>

City Council Agenda Item Request

Date: 10/11/2023

Name: Jeff Lampi

Department: Water Department

Item: Engineering Services for DWSRF - ARPA-2 Project

Meeting date requested: 10/19/2023

Explanation for request:

Administration is requesting Council's approval to retain C2AE from Escanaba, MI, for engineering services with regard to the City of Escanaba DWSRF ARPA-2 Project in an amount not to exceed \$2,320,900.00.

Engineering costs can be reimbursable to the City as the project progresses throughout design and construction.

***once council approves this expendature, C2AE will be engaged to act upon phase I for prelimnary engineering and additional engineering at a cost of \$556,700.00.



October 11, 2023

Mr. Jeff Lampi
Utilities Superintendent
City of Escanaba
P.O. Box 948
Escanaba, MI 49829

Re: Proposal for Professional Services: DWSRF ARPA Grant Funded Watermain and LSLR Project – 2024 Phase II

Dear Mr. Lampi,

Congratulations to the City of Escanaba for again being selected as a recipient of another water infrastructure grant!

The City of Escanaba has been awarded an American Rescue Plan Act (ARPA) Grant through EGLE DWSRF for Lead Service Line and Watermain Replacements throughout the City. The project total is an astonishing \$20 Million with 100 percent grant and no loan component. This is the second grant of this magnitude the City has been awarded.

Project Understanding

For the project to proceed, the next step will be the development of Preliminary Engineering Plans for the LSLR Project as well as the Watermain Project. This will involve preparation of the Basis of Design, preliminary/final drawings, outline specifications and written descriptions of the project. In addition, soil borings, on-site investigations, and control surveys, will be needed in preparation of these documents.

The ARPA Grant project allows for time variations in the design phase of the project. However, C2AE proposes to maintain a design schedule that is consistent with EGLE's SRF 4th Quarter financing schedule. As such, preliminary plan and specification submittal is scheduled for mid-February 2024. To meet this timeline, C2AE and the City will need to move forward with Preliminary Engineering Phase quickly.

Scope

In our efforts to secure the maximum amount of ARPA grant funding, both the Lead Service Line Replacement Project and the Watermain Replacement Project need to be bid out and under construction in 2024. Fortunately, C2AE has been working with the City for several years on the water system design. We have the background mapping, specifications, and tools to assist with plan development for this next phase of work. In addition, several of our staff engineers are familiar with the unique design and construction techniques associated with service line replacement.

The Watermain Project, as outlined in the Project Plan, requires full design and engineering for the preparation of construction drawings and specifications. C2AE has background mapping for the City and can begin the design work immediately to meet the project schedule. The coordination efforts between the City staff and C2AE (on Phase I ARPA Project) allows ARPA Phase II watermain replacements to progress at an accelerated pace.



Overflow pipe and miscellaneous rehabilitation items of the south elevated water tank are included in the ARPA funding. C2AE has included Dixon Engineering as a sub consultant in our Engineering Proposal for the south tower rehabilitation design work. Dixon has provided the City with previous tower studies and are very familiar with the rehabilitation needs on the tank.

We have attached the EJCDC Agreement for Professional Services prepared for the LSLR and Watermain Projects. C2AE is requesting the City of Escanaba authorize C2AE to proceed with the following Engineering Services line items to keep in sequence with financing schedules. Basic Services and Additional Engineering Services can be found in Exhibit C of the Agreement. Engineering costs can be reimbursable to the City once the project grant financing has been completed through EGLE.

Fee

Breakdown of engineering task can be found in Exhibit C of the Agreement. A summary of costs for the project is as follows:

	LSLR Project	Water Main Project	
Preliminary Engineering Phase: 🗡	\$ 45,000	\$225,000	270,000
Final Design Phase:	\$ 40,000	\$ 195,000	235,000
Bidding and Negotiating Phase:	\$ 15,000	\$ 34,000	49,000
Construction Engineering Phase	\$ 135,000	\$ 219,000	354,000
Post Construction Phase:	\$ 17,000	\$ 40,000	57,000
RPR Field Services:	\$502,700	\$566,500	1,069,200
Additional Engineering Services:**	\$ 76,000	\$ 210,700	286,700
	\$830,700 Total	\$1,490,200 Total	\$ 2,320,900

The Engineering Costs are based on two potential construction contracts for LSLR and Watermain Project work. We are proposing two Resident Project Representative/Inspectors (RPR's) required for this project and an assistant RPR to support the multiple contractor operations on the overall project due to the magnitude of work to be performed.

C2AE is requesting approval to proceed with design engineering and additional services needed to complete work through the bidding phase of the project. C2AE will request future approvals to proceed with the construction engineering services once the project has been successfully bid out to contractors.

C2AE thanks the City of Escanaba for the opportunity to work in our hometown and build a better community. We look forward to working with you and your staff on this project.

Sincerely, C2AE	* Phase I = \$556,700	Accepted by:	
Darren Pionk, PE Project Manager		City of Escanaba	
		Date	

Agenda Item: <u>NB-4</u>
Date: 10-19-2023

City Council Agenda Item Request

Date: 10/12/2023

Name: Jeff Lampi

Department: Water Department

Item: Engineering Services for WTP Filter Bed Rehabilitation

Meeting date requested: 10/19/2023

Explanation for request:

Administration is requesting Council's approval to retain C2AE from Escanaba, MI, for engineering services with regard to the City of Escanaba Water Treatment Plant Filter Bed Rehabilitation in an amount not to exceed \$8,000.00.

Money is not budgeted, but is avialable in the water fund under professional services for this type of work.



October 11, 2023

Mr. Jeff Lampi Utilities Superintendent City of Escanaba P.O. Box 948 Escanaba, MI 49829

Re: Proposal for Professional Services: City of Escanaba WTP Filter Bed Rehabilitation

Dear Jeff,

Thank you very much for the opportunity to submit this proposal for engineering services for the design of the Water Treatment Plant Filter Bed Rehabilitation. C2AE looks forward to assisting the City in this necessary rehabilitation project at the plant.

Project Understanding

C2AE understands the Water Treatment Facility currently has four separate water filters. One filter bed is currently experiencing filter media loss with holes developing under the sand. The repair work is critical to the operation of the plant and needs attention in the near future. One other filter was previously rehabilitated by the City, and the final two filters will be planned for rehabilitation in the upcoming Invitation To Apply (ITA) for potential EGLE funding.

Per our discussion, our Process Engineer, Michael Faeth, will oversee the design approach for the filter beds. His previous experience with filter bed rehabilitation and vendor knowledge of underdrain systems will ensure that the project meets the City's expectations.

Scope

C2AE proposes to provide services for the Filter Bed Replacement project as follows:

Design Phase Services

- 1. Project Initiation Meeting: Conduct a project initiation meeting with the City to review project parameters, scope of services and project schedule contained in the project management plan. Review any previous schematic plans or studies with the City.
- 2. Base map Preparation: Develop base drawings of the current water plant filter room and relevant process pipe works. Investigate options of filter bed underdrain systems and present findings to the City for approval.
- 3. Plan Drawings: Prepare plan drawings which will show the placement of all improvements as well as construction notes. Plan drawings shall include:
 - a. Title Sheet
 - b. General Notes and Details



- c. Existing Conditions Plan
- d. Filter Bed Demolition Plan
- e. Filter Bed Improvement Plan
- 4. Technical Specifications: Prepare technical specifications to establish quality of all materials, equipment, installation, and construction.
- 5. Interim Meetings: Meet with the City as required throughout the project to review preliminary drawings, costs, and review progress. Provide written progress reports.
- 6. Permit Applications: Complete applications to the Michigan Department of Environment, Great Lakes, and Energy (EGLE) for execution by the City, as required for the project.
- 7. Final Revisions: Make final revisions based on the above review and comment.

Bidding Phase Services

No bidding phase services are included, but can be added at a later date if desired.

Construction Phase Services

No Construction Engineering phase services are included, but can be added at a later date if desired.

Assumptions

- 1. The City will provide any as-built drawings of plant that are not currently in possession of C2AE.
- 2. The City will provide the front end manual and bid the project. C2AE to provide technical specifications only.
- 3. The City will perform construction administration.

Schedule

Design phase services will commence upon contract approval and be completed within 90 days. Bidding to occur in the Winter of 2023/2024. Construction is anticipated to occur during the 2024 construction season and be completed by July 31, 2024.

Fee

C2AE proposes to provide design phase services for the project on a lump sum basis for \$7,500.00.





We have included our Standard Contract Provisions as part of our proposal. If the terms and conditions as stated are acceptable, please countersign and return one (1) copy to our office. Please do not hesitate to contact us should you have any questions or concerns, or if you need additional information.

Sincerely, C2AE	Accepted by:	
Darren Pionk, PE Project Manager	City of Escanaba	
Roger Marks, PE Client Services Leader	Date	

Agenda Item: <u>NB-5</u>
Date: <u>10-19-203</u>

City Council Agenda Item Request

Date: 10/11/2023

Name: Jeff Lampi

Department: Wastewater

Item: Ludington Lift Station Inspection - Asbestos

Meeting date requested: 10/19/2023

Explanation for request:

Administration is requesting city council's approval to obtain the services of Pearson Asbestos Abatement of Escanaba, Michigan to carry out non-destructive inspection of the Ludington Lift Station building for asbestos containing materials and paint for lead and cadmium in an amount not to exceed \$2,000.00.

Money is budgeted and available for this type of activity.

PEARSON ASBESTOS ABATEMENT



A division of OHA, LLC. 2715 Danforth Road ~ P.O. Box 1292 Escanaba, MI 49829

Phone: 906-786-3001 ~ Fax: 906-789-4010

Email: admin@pearsonasb.com

October 6, 2023

City of Escanaba Waste Water Treatment Plant P.O. Box 948 1900 Willow Creek Road Escanaba, MI 49829 Phone: 906-786-1301

Email: ilampi@escanaba.org

Attn: Jeff Lampi

RE: Ludington Lift Station Inspection

To have a State of Michigan accredited asbestos inspector carry out a non-destructive inspection of the Ludington Lift Station building for asbestos containing building materials and paint for lead and cadmium. All visible suspect materials will be sampled. We will not be busting holes in walls and ceilings, busting up floors, etc. unless specifically instructed to do so.

Inspection fee:		\$450.00
Lab fees:		
Sample analysis for asbestos per layer		
of material that the lab analyzes (24 hr.)	\$28.00 ea.	\$280 - \$560*
Paint samples for lead/cadmium as		
required by the State of Michigan	\$50.00 ea.	\$250 - \$500*
Total:		\$980 - \$1,510

*We anticipate ending up with approximately 10 - 20 layers of asbestos materials sampled. And 5 - 10 lead/cadmium samples.

Please note: This is a ballpark price. Depending on the total number of layers of materials, the cost could be less or could exceed the ballpark price.

Thank you for the opportunity to quote. If you have any questions or would like to schedule this job, please call.

PEARSON ASBESTOS ABATEMENT

Rob Arndt

Rob Arndt General Manager RA/lk

Acceptance of the above work		
	Signature	Date

TERMS: Balance due within 30 days after invoicing. Finance Charges 1.5%. Please sign this and return. Thank you. This proposal is good for 30 days unless otherwise stated.

Agenda Item: NB-6
Date: 10-19-2023

City Council Agenda Item Request

Date: 10/13/2023

Name: James McNeil

Department: Manager

Item: Resolution - Termination of Participation in MERS Defined Benefit Plan

Meeting date requested: 10/19/2023

Explanation for request:

Administration is requesting City Council approval of a resolution terminating participation in the MERS Defined Benefit Plan. This resolution is in accordance with the MERS Plan Termination Policy. Assets, liabilities, and fiduciary duties will transfer to the successor plan.

Resolution 23-22

Resolution by the City of Escanaba

Regarding Termination of Participation in MERS' Defined Benefit Plan

Whereas, on October 19, 2023, the City Council of the City of Escanaba elected, by two-thirds vote of its membership, to terminate participation for all divisions of the MERS Defined Benefit Plan and transfer assets, liabilities and fiduciary duties to the City of Escanaba Defined Benefit Retirement Plan ("Successor Plan"), a governmental pension plan qualified under Section 401(a) of the Internal Revenue Code ("IRC");

Whereas, MERS Plan Document Section 11 and MERS' Termination Policy and Procedure require that the City Council certify and affirm certain information as a condition of termination of participation in the MERS Defined Benefit Plan; and

Whereas, the City Council has reviewed MERS Plan Document Section 11 and MERS' Termination Policy and Procedure and affirms it satisfies all requirements and conditions for termination of participation from the MERS Defined Benefit Plan;

Now, therefore, the City Council hereby resolves:

- 1. Termination of participation in the MERS Defined Benefit will not result in a diminishment of any accrued financial benefit by members, deferred vested former members, retirees or beneficiaries, as defined under the MERS Plan Document, under Article 9, Section 24 of the Michigan Constitution;
- 2. The City of Escanaba acknowledges the requirement to fund the Successor Plan for the benefit of its participants, as required by all applicable laws and regulations and agrees to so fund the plan;
- 3. The City of Escanaba and its designees will be the fiduciary for the Successor Plan to administer the Successor Plan, receive and invest the assets and liabilities to be transferred from the MERS Defined Benefit Plan to the Successor Plan, as set forth in the termination valuation conducted by MERS.
- 4. The Successor Plan is a qualified plan under IRC section 401(a); it has, to the best of City Council's knowledge, maintained that qualified status through the date of this Agreement, and intends to maintain such status until final plan termination;
- 5. As the plan sponsor of the Successor Plan, City of Escanaba has adopted or has directed the adoption of reasonable actuarial assumptions for the Successor Plan, developed using the Actuarial Standards of Practice of the Actuarial Standards Board.

- 6. On the effective date of termination of participation in the MERS Defined Benefit Plan, to be determined by MERS, all responsibilities and liabilities of investment fiduciaries as set forth under the Public Employees Retirement Investment Security Act, Act 314 of 1965, as amended, and fiduciary, plan administrator and trustee (under all other applicable state or federal common or statutory laws or regulations) will transfer from MERS to the Successor Plan and its Fiduciary Officials as defined below; and
- 7. In accordance with MERS Plan Document Section 11 and the MERS' Termination Policy and Procedure, the City Council directs the Fiduciary Officials for the Successor Plan to execute this resolution as an acknowledgment of the fiduciary duties being transferred.

So resolved on October 19, 2023, at a regularly scheduled meeting of City Council.
Authorized Designee of City Council of the City of Escanaba
Authorized Signature:
Title:
Fiduciary Officials Acknowledgment:
For City of Escanaba
Signature:
Name:
Title:

Agenda Item: NB-7
Date: 10-19-2023

City Council Agenda Item Request

Date: 10/13/2023

Name: James McNeil

Department: Manager

Item: Resolution - Adoption of City of Escanaba Defined Benefit Retirement Plan

Meeting date requested: 10/19/2023

Explanation for request:

Administration is requesting City Council approval of a resolution adopting all the components of the City of Escanaba Defined Benefit Retirement Plan. This approval includes the Plan document, establishment of the Retirement Board, Board Rules of Procedure, and Statement of Investment Policy,

Resolution 23-23

CITY OF ESCANABA CITY COUNCIL

RESOLUTION ADOPTING THE CITY OF ESCANABA'S DEFINED BENEFIT RETIREMENT PLAN AND ESTABLISHMENT OF RETIREMENT BOARD

At a regular meeting of the City Council for the City of Escanaba, Delta County, Michigan,
held at City Hall on the day of, 2023, at: p.m.
PRESENT:
ABSENT:
The following preamble and resolution were offered by
and seconded by

WHEREAS, the City currently participates in the Municipal Employees' Retirement System of Michigan ("MERS") Defined Benefit Plan (the "MERS Plan"). The MERS Plan is administered by MERS, a professional retirement services company that was created to administer retirement plans for Michigan's local units of government;

WHEREAS, the lack of funding progress of the MERS Plan has contributed to financial stress on the City that has intensified in recent years. City management has lost confidence in MERS and believes that it could achieve better financial outcomes by terminating participation in the MERS Defined Benefit Plan and establishing a successor plan that would be overseen by a retirement board established by the City;

WHEREAS, pursuant to the MERS Retirement Board Termination Policy and Procedure, this City Council voted by resolution to terminate participation in the MERS Defined Benefit Plan and to transfer all of its MERS Plan assets and liabilities to a successor defined benefit plan.

NOW, THEREFORE, BE IT RESOLVED BY the City Council of City of Escanaba that:

1. The City Council adopts the City of Escanaba Defined Benefit Retirement Plan (the "Plan"), a copy of which is attached hereto.

- 2. The City Council approves the transfer of assets and liabilities from the MERS Defined Benefit Plan to the Plan.
- 3. The City Council adopts the City of Escanaba Defined Benefit Retirement Plan Statement of Investment Policy, a copy of which is attached hereto.
- 4. The City Council establishes the City of Escanaba Defined Benefit Retirement Plan Board (the "Retirement Board"). The Retirement Board will be a five-member board consisting of the City Manager, the City Controller, the City Treasurer, one member of the City Council, and one member of the public with preference to any Plan member. The Retirement Board is responsible for oversight and decision making relating to the general operation of the Plan in accordance with the Statement of Investment Policy.
- 5. The City Council adopts the City of Escanaba Defined Benefit Retirement Plan Board Rules of Procedure, a copy of which is attached hereto.
- 6. The City Manager is authorized to execute the Plan document and any other documents necessary to implement the foregoing Resolutions.

A vote on the above Resolution was taken and was as follows:

ADOPTED:					
YEAS:					
NAYS:					
STATE OF MICHIGAN)				
COUNTY OF DELTA) ss.)				
I, the undersigned, the duly qual CERTIFY that the foregoing is a Council of said City at a regular	true and complet	te copy of ce	ertain proceed	ings taken by	the City
		– Pi	hil DeMay – C	City Clerk	

CITY OF ESCANABA DEFINED BENEFIT RETIREMENT PLAN

TABLE OF CONTENTS

ARTICLE I	- Purpose	1
ARTICLE II	- Definitions	1
ARTICLE II	I - Participation and Service	6
3.1	Eligibility	
3.2	Credited Service	6
ARTICLE I	V - Eligibility for Benefits	7
4.1	Normal Retirement and Late Retirement	7
4.2	Eligibility for Reduced Early Retirement Benefit	7
4.3	Eligibility for Unreduced Early Retirement Benefit	8
4.4	Eligibility for Deferred Vested Pension	8
4.5	Eligibility for Death Benefit	8
4.6	Eligibility of Certain Individuals for Disability Retirement Benefit	8
ARTICLE V	- Amount and Form of Benefits	8
5.1	Normal Retirement Benefit	8
5.2	Reduced Early Retirement Benefit	9
5.3	Unreduced Early Retirement Benefit	
5.4	Late Retirement Benefit	9
5.5	Death Benefit	9
5.6	Disability Benefit	15
5.7	Deferred Vested Pension Benefit	17
5.8	Payment of Benefit	
5.9	Maximum Annual Pension Benefit by Law	
5.10	Distribution Restrictions	
5.11	Cost of Living Adjustment	34
5.12	Employment Termination Prior to Effective Date	34
ARTICLE V	T - Payments from the Trust Fund	35
6.1	Pension Benefit Payments	
6.2	Death Benefit Payment	
6.3	Reemployed Retired Participant	
6.4	Non-Duplication of Benefits	
6.5	Benefits Payable to Incompetents	35
6.6	Nonalienation of Benefits	35
ARTICLE V	II - Financing	36
7.1	Trust Fund	
7.2	Employer Contributions	
7.3	Plan-to-Plan Transfers from Qualified Plans	
ARTICLE V	TIII - Administration of the Plan	36
8.1	Plan Administrator	
8.2	Responsibilities	
8.3	Records and Reports	
8.4	Other Plan Administrator Powers and Duties	
8.5	Rules and Decisions	

8.6	Authorization of Benefit Payments	39
8.7	Application and Forms for Pension	
8.8	Facility of Payment	
8.9	Indemnification	
ARTICLE IX	- Amendment and Termination of Plan	39
9.1	Right to Terminate	39
9.2	Right to Amend	
9.3	Effect of Amendment or Termination	39
9.4	Allocation of the Fund on Termination or Partial Termination	40
9.5	Disposition of Assets on Termination or Partial Termination	40
9.6	Residual Amounts	
9.7	No Additional Liability	40
9.8	Successor Sponsor	40
9.9	Plan Assets	41
ARTICLE X	- Miscellaneous	41
10.1	Unclaimed Pension Checks	41
10.2	Correction of Errors	41
10.3	Nonguarantee of Employment	41
10.4	Gender and Number	

CITY OF ESCANABA DEFINED BENEFIT RETIREMENT PLAN

ARTICLE I - PURPOSE

The City of Escanaba hereby adopts the City of Escanaba Defined Benefit Retirement Plan ("Plan"), effective as of the Effective Date listed in Section 2.12. Prior to the Effective Date, the City of Escanaba participated in the Municipal Employees' Retirement System of Michigan Retirement Plan (the "Prior Plan"). The City elected to terminate participation in the Prior Plan as of the Effective Date, and to transfer all of its assets and liabilities from the Prior Plan to this Plan effective as of the Transfer Date listed in Section 2.28. The rights and benefits, if any, of an Employee who terminated employment prior to the Effective Date shall be determined in accordance with the provisions of the Prior Plan in effect as of the date on which the Employee terminated employment unless this Plan provides otherwise.

The Plan is intended to meet the requirements of Sections 401(a) and 501(a) of the Internal Revenue Code of 1986, as amended.

ARTICLE II - DEFINITIONS

- 2.1 <u>Accrued Benefit</u> means the benefit commencing at the Participant's Normal Retirement Date, determined in accordance with Section 5.1. In the event a Participant terminates employment prior to his or her Normal Retirement Date or Early Retirement Date, the Participant's Accrued Benefit shall be equal to the amount determined in accordance with Section 5.1, computed as of the Participant's date of employment termination based on Final Average Compensation, Compensation and Credited Service determined as of the date of such termination.
- 2.2 <u>Actuarial or Actuarially Equivalent</u> means equality in value of the aggregate amounts expected to be received under different forms of payment. Such equivalents shall be based on actuarial assumptions approved from time to time by the Plan Administrator with the advice of the Plan actuary.

(a) Plan Assumptions.

- (1) <u>Benefit Commencement Dates Preceding the Transfer Date.</u> With respect to any Participant whose monthly Plan benefit payments commence after the Effective but before the Transfer Date, Actuarial Equivalence among optional forms of benefit shall be calculated using (i) an interest rate of 7.00% per annum; and (ii) the mortality assumptions that are described in Section V.4.a. of the Municipal Employees' Retirement System Actuarial Policy dated June 16, 2023.
- (2) <u>Benefit Commencement Dates On or After the Transfer Date.</u> With respect to any Participant whose monthly Plan benefit payments commence on or after the Transfer Date, Actuarial Equivalence among optional forms of benefit shall be calculated using interest rate and mortality assumptions determined by the Plan actuary as set forth in a separate actuarial policy.
- (b) <u>Section 415 Assumptions</u>. Special rules apply to the calculation of Actuarially Equivalent benefits for purposes of applying the annual limits on benefits contained in

- Section 5.9. Those special rules are contained in Section 5.9. The definitions that are set forth below shall apply for purposes of Section 5.9.
- (1) For Calendar Years 2003 Through 2007. This subsection (b)(1) applies for calendar years that begin on or after January 1, 2003 and before January 1, 2008. The "Applicable Mortality Table" means the mortality table set forth in Rev. Rul. 2001-62, 2001-53, I.R.B. 632 and the "Applicable Interest Rate" means the interest rate set forth in Section 417(e)(3)(A)(ii)(II) of the Code for the first calendar month preceding the calendar month during which the Annuity Starting date occurs.
- (2) <u>For Calendar Year 2008 and Subsequent Calendar Years</u>. This subsection (b)(2) applies for calendar years that begin on or after January 1, 2008.
- [a] The "Applicable Mortality Table" means the mortality table prescribed by Code Section 417(e)(3). For any distribution with an annuity starting date on or after the effective date of these subsections and before the adoption date of these subsections, if application of the amendment as of the annuity starting date would have caused a reduction in the amount of any distribution, such reduction is not reflected in any payments made before the adoption date of these subsections. However, the amount of any such reduction that is required under Code Section 415(b)(2)(B) must be reflected actuarially over any remaining payments to the Participant.
- [b] The "Applicable Interest Rate" means the rate of interest determined by the applicable interest rate described by Code Section 417(e) after its amendment by PPA. Specifically, the applicable interest rate shall be the adjusted first, second, and third segment rates applied under the rules similar to the rules of Code Section 430(h)(2)(C) determined as of the first calendar month preceding the first day of the calendar month during which the annuity starting date occurs. For this purpose, the first, second, and third segment rates are the first, second, and third segment rates which would be determined under Code Section 430(h)(2)(C) if:
- [i] Code Section 430(h)(2)(D) were applied by substituting the average yields for the month described in the preceding paragraph for the average yields for the 24-month period described in such Section, and
- [ii] Code Section 430(h)(2)(G)(i)(II) were applied by substituting "Section 417(e)(3)(A)(ii)(II) for "Section 412(b)(5)(B)(ii)(II)," and
- [iii] the applicable percentage under Code Section 430(h)(2)(G) is treated as being 20% in 2008, 40% in 2009, 60% in 2010, and 80% in 2011.

For purposes of this subsection (b), a Participant's annuity starting date shall be the first day of the first period for which an amount is paid as an annuity or, in the case of a benefit that is not payable in the form of an annuity, the first day on which all events have occurred which entitled to Participant to such benefit.

2.3 <u>Code</u> means the Internal Revenue Code of 1986, as amended from time-to-time.

- 2.4 <u>Compensation</u> shall have the same meaning as such term had under the Prior Plan as of the date immediately preceding the Effective Date, and as further described on Page 5 of the MERS Defined Benefit Plan Adoption Agreement Addendum for the Prior Plan (incorporated by reference to this Plan as Attachment A).
- Certain Qualifying Military Service. A Participant who terminates employment with the Employer because of qualified military service in the uniformed services of the United States, and is reemployed by the same Employer after completion of the qualified military service, shall be treated as receiving compensation from the Employer during such period of qualified military service equal to (i) the compensation the Participant would have received during such period if the Participant were not in qualified military service, determined based on the rate of pay the Participant would have received from the Employer but for the absence during the period of qualified military service, or (ii) if the compensation the Participant would have received during such period was not reasonably certain, the Participant's average compensation from the Employer during the twelve (12) month period immediately preceding the qualified military service (or, if shorter, the period of employment immediately preceding the qualified military service) within the time limits specified under USERRA. If a Participant or former Participant dies while performing qualified military service, the Participant's or former Participant's beneficiary shall be entitled to any benefits (including Employer Contributions relating to the period of qualified military service), and the rights and features associated with those benefits, that would have been provided if the Participant or former Participant had been reemployed by the Employer and died while an active eligible Participant.
- (b) <u>Compensation Limit</u>. The Compensation of each Participant taken into account in determining benefit accruals in any Plan Year beginning after December 31, 2022, shall not exceed \$265,000 (or such other amount provided in the Code). Such amount shall be adjusted for increases in the cost of living in accordance with Code Section 401(a)(17)(B), except that the dollar increase in effect on January 1 of any calendar year shall be effective for the determination period beginning with or within such calendar year. For any short determination period, the Compensation limit shall be an amount equal to the Compensation limit for the calendar year in which the determination period begins multiplied by the ratio obtained by dividing the number of full months in the short determination period by twelve (12).
- 2.5 <u>Teamsters Participant</u> means a Participant who was a member of the Teamsters Division of the Prior Plan as of the date immediately preceding the Effective Date.
- 2.6 <u>General Electric Participant</u> means a Participant who was a member of the General Elect Division of the Prior Plan as of the date immediately preceding the Effective Date.
- 2.7 <u>Non-Union Participant</u> means a Participant who was a member of the Non Union Division of the Prior Plan as of the date immediately preceding the Effective Date.
- 2.8 <u>Dispatchers Participant</u> means a Participant who was a member of the Dispatchers Division of the Prior Plan as of the date immediately preceding the Effective Date.
- 2.9 <u>Part-Time Emp Participant</u> means a Participant who was a member of the Part-Time Emp Division of the Prior Plan as of the date immediately preceding the Effective Date.

- 2.10 <u>Water/Wastewater Teamsters Participant</u> means a Participant who was a member of the Water/Wastewater Teamsters Division of the Prior Plan as of the date immediately preceding the Effective Date.
- 2.11 <u>Early Retirement Date</u>. A Participant's Early Retirement Date is the first day of the month coinciding with or next following the earlier of: (1) the date on which a Participant has attained age 50 and completed at least 25 Years of Service for vesting with the Employer; or (2) the date on which a Participant has attained age 55 and completed at least 15 Years of Service for vesting with the Employer.
- 2.12 <u>Effective Date</u> of this Plan shall be October 19, 2023, unless otherwise indicated herein as to specific Plan provisions.
- 2.13 <u>Eligible Spouse or Spouse</u> means a spouse to whom the Participant is legally married on the Participant's benefit commencement date; provided, however, that in the event of a Participant's death before the Participant's benefit commencement date, Eligible Spouse shall mean the spouse to whom the Participant has been married throughout the one-year period immediately preceding the death of the Participant.
- 2.14 <u>Employee</u> means any person who is receiving remuneration for personal services rendered to the Employer or to any other employer required to be aggregated with the Employer under Code Sections 414(b), (c), (m) or (o) (an "Affiliated Employer") (or who would be receiving such remuneration except for an unauthorized leave of absence). Notwithstanding the foregoing, Employee shall not include:
- (a) any employee of an Affiliated Employer, unless such Affiliated Employer has specifically adopted this Plan in writing;
- (b) any individual who is not reported on the payroll records of the Employer as a common law employee (in particular, it is expressly intended that any individual who us not treated as common law employee by the Employer on its payroll records and any out-sourced worker shall not be an Employee and shall be excluded from Plan participation, even if a court or administrative agency determines that such individual is a common law employee and not an independent contractor);
- (c) any individual who is classified under the Employer's regular personnel policies as an intern or seasonal employee; and
- (d) any individual who voluntarily elects in writing or by employment contract to waive participation in the Plan, provided each such individual's basic rate of compensation from the Employer exceeds the maximum earnings covered by Social Security. Such waiver of participation shall be irrevocable and must comply with relevant law.

Any individual who was excluded from the Plan on the basis that he or she was not a common law employee of the Employer and who is later found by a court of law to be a common law employee of the Employer shall, except in the case of an individual who is described in subsection (b) above be eligible to participate in the Plan only from the date on which the court's order finding the individual to be a common law employee is entered.

- 2.15 <u>Employer</u> means the City of Escanaba, any successor that shall maintain this Plan and any predecessor that has maintained this Plan.
- 2.16 <u>Fiduciary (or Fiduciaries)</u> means the City of Escanaba is the Named Fiduciary. Other persons or entities may become Plan fiduciaries to the extent of any delegation of discretion under the Plan by the Named Fiduciary that is accepted by said other person or entity.
- 2.17 <u>Final Average Compensation</u> means one-third (1/3) of the aggregate amount of compensation paid and earned by a Participant during the period of three (3) consecutive years (determined using whole calendar months) of the Participant's credited service in which the aggregate amount of Compensation paid and earned is highest. If the Participant has less than three (3) years of credited service, Final Average Compensation means the aggregate amount of compensation paid and earned by the Participant divided by the Participant's credited service. For purposes of determining Final Average Compensation, any portion of a lump sum payment earned for work performed outside the Final Average Compensation period shall be excluded.
- 2.18 <u>Hour of Service</u>. means each hour for which an Employee is paid, or entitled to payment, for the performance of duties for the Employer.
- 2.19 <u>Normal Retirement Date</u> means the later of (1) the date on which the Participant attains age 60, or (2) the Participant's tenth anniversary of participation in the Plan.
- 2.20 <u>Participant</u> means an Employee who is participating in the Plan under the terms of Section 3.1.
- 2.21 Plan means the City of Escanaba Defined Benefit Retirement Plan as set forth herein.
- 2.22 <u>Plan Administrator</u> means the City of Escanaba.
- 2.23 <u>Plan Year</u>. The initial Plan Year shall be the period commencing on the Effective Date and ending on June 30, 2024. Thereafter, the Plan Year shall be the 12-consecutive month period commencing on July 1 and ending on June 30 of each year.
- 2.24 <u>Prior Plan</u> means the Municipal Employees' Retirement System of Michigan Defined Benefit Plan as in effect on the date immediately preceding the Effective Date.
- 2.25 Retirement means the date as of which a Participant has terminated employment for reasons other than death and after the Participant has fulfilled all requirements to be eligible for a normal, early or disability retirement benefit. Retirement shall occur on the day that immediately follows the Participant's last day of employment (or authorized leave of absence, if later). For this purpose an authorized leave of absence means an absence that is authorized by the Employer under a consistent and non-discriminatory application of the Employer's standard personnel practices, provided that the Participant retires or returns within the period specified in the authorized leave of absence.
- 2.26 <u>Straight Life Pension</u> means a pension payable for the life of the Participant, without provision for payment of benefits after his or her death.
- 2.27 Sponsor means the City of Escanaba.

- 2.28 <u>Transfer Date</u> means the date on which plan assets will be transferred from MERS to this Plan pursuant to the MERS Retirement Board Termination Policy and Procedure.
- 2.29 <u>Trust (or Trust Fund or Fund)</u> means the fund maintained for the benefit of Plan Participants pursuant to the City of Escanaba Defined Benefit Retirement Plan Trust Agreement as from time to time amended.
- 2.30 <u>Trustee</u> means the person or entity named as trustee herein or in any separate trust forming a part of this Plan, and any successors.
- 2.31 <u>Unreduced Early Retirement Date.</u> A Participant's Unreduced Early Retirement Date is the first day of the month coinciding with or next following the date on which a Participant has attained age 55 and completed at least 25 Years of Service for vesting with the Employer.

ARTICLE III - PARTICIPATION AND SERVICE

- 3.1 <u>Eligibility</u>. Any Employee, former Employee, or beneficiary of a former Employee (1) who participated in the Prior Plan on the date immediately preceding the Effective Date; and (2) for whom an asset and related liability are transferred to this Plan from the Prior Plan on the Transfer Date became a Participant in this Plan as of the Effective Date.
- 3.2 <u>Credited Service</u>. The following rules apply for purposes of determining a Participant's credited service under the Plan.
- (a) General Rule. Prior service to which a Participant is entitled, including periods during which a Participant is in receipt of workers' compensation weekly benefits, shall be credited to the Participant's individual service account. Service shall be credited in years and twelfths (1/12) of a year. Not more than one year of credited service shall be credited to a Participant on account of all service rendered to the Employer in any period of 12 consecutive months. Not more than 1/12 of a year of credited service shall be credited to a Participant on account of all service rendered to the Employer in a calendar month. Credited service shall not be credited for any calendar month during which a Participant completes less than 20 Hours of Service for the applicable month.
- (b) <u>Special Rule.</u> The Special Rule described in this subsection 3.2(b) shall apply solely for purposes of determining service credit for purposes of vesting and early retirement eligibility, and shall not apply for benefit accrual purposes.
- (1) <u>Act 88 Service Credit.</u> For purposes of vesting and early retirement eligibility, a Participant shall be credited with service performed for another "reciprocal unit", as that term is defined in MCL 38.1102(d); provided, however, that such service credit will only include service performed for the reciprocal unit during the period of time during which the reciprocal unit elected to be subject to Michigan Public Act 88 of 1961.
- (2) <u>Prior MERS Service Credit.</u> For purposes of vesting and early retirement eligibility, a Participant shall be credited with service performed before the Effective Date for any municipality or court that participated in the MERS Defined Benefit Plan or the MERS Defined Contribution Plan while the service was performed (a "MERS Participating Employer"); provided, however, that the following service will not be credited under this subsection:

- (i) Service of less than one year with the MERS Participating Employer;
- (ii) Service that has been forfeited under Section 42 of the MERS Defined Benefit Plan Document, except that non-vested service related to employee contributions that are distributed under the required minimum distributions provisions of Code Section 401(a)(9) shall be recognized;
- (iii) Service that preceded a break in membership of more than 240 consecutive months;
- (iv) Service concurrently acquired in more than one MERS Participating Employer;
- (v) Non-vested service accrued at a participating municipality or court before the municipality or court became a MERS Participating Employer; and
- (vi) Service performed for a municipality or court after the municipality or court has terminated participation with MERS.
- 3.3 <u>Purchased Service</u>. A Participant may purchase up to five years of credited service if (1) the purchase is permitted by and consistent with a resolution adopted by the City of Escanaba City Council; and (2) the Participant makes an immediate contribution to the Plan equal to an amount that is Actuarially Equivalent to the present value of the Participant's accrued benefit after crediting the specified amount of purchased service. Any service credit purchased under this Section must not result in the Participant exceeding any applicable benefit accrual or payment limitation imposed by this Plan or relevant law.

ARTICLE IV - ELIGIBILITY FOR BENEFITS

4.1 <u>Normal Retirement and Late Retirement</u>. Each eligible Participant shall receive a normal or late retirement benefit calculated pursuant to Sections 5.1 and 5.4, respectively. Payment of said normal or late retirement benefit shall commence as of the first day of the month coincident with or next following the later of: (a) the date on which the Participant has terminated employment with the Employer and has attained his or her Normal Retirement Date; or (b) the Participant's Late Retirement Date.

A Participant's Late Retirement Date shall be the date after his or her Normal Retirement Date on which he or she has terminated employment with the Employer.

4.2 <u>Eligibility for Reduced Early Retirement Benefit</u>. A Participant whose employment with the Employer is terminated on or after his or her Early Retirement Date shall be entitled to a reduced early retirement benefit as provided in Section 5.2 of the Plan. Said early retirement benefit shall commence as of the first day of the month coinciding with or next following the later of: (1) the date on which such Participant has terminated employment with the Employer; or (2) the date on which the Participant's properly completed application for an early retirement benefit is approved by the Plan Administrator.

- 4.3 <u>Eligibility for Unreduced Early Retirement Benefit</u>. A Participant whose employment with the Employer is terminated on or after his or her Unreduced Early Retirement Date shall be entitled to an unreduced early retirement benefit as provided in Section 5.3 of the Plan. Said unreduced early retirement benefit shall commence as of the first day of the month coinciding with or next following the later of: (1) the date on which such Participant has terminated employment with the Employer; or (2) the date on which the Participant's properly completed application for an early retirement benefit is approved by the Plan Administrator
- 4.4 <u>Eligibility for Deferred Vested Pension</u>. A Participant whose employment with the Employer is terminated prior to his or her Normal Retirement Date shall be entitled to a deferred vested pension, commencing not earlier than his or her Normal Retirement Date; provided, however, that at the election of the Participant, said deferred vested retirement benefit shall commence as of the first day of any month coinciding with or next following his or her Early Retirement Date.
- 4.5 <u>Eligibility for Death Benefit</u>. The Eligible Spouse or other designated beneficiary of a Participant who dies may be entitled to a death benefit as determined under, and commencing in accordance with, Section 5.5 of the Plan.
- 4.6 <u>Eligibility of Certain Individuals for Disability Retirement Benefit</u>. A Participant may be entitled to a disability benefit as determined under, and commencing in accordance with, Section 5.6 of the Plan.

ARTICLE V - AMOUNT AND FORM OF BENEFITS

- 5.1 <u>Normal Retirement Benefit</u>. Each Participant whose benefit commencement date occurs after the Effective Date and who is eligible to receive a normal retirement benefit in accordance with Section 4.1 shall be entitled to a Straight Life Pension that is calculated as set forth below.
- (a) <u>All Participants Except for Part-Time Emp Participants.</u> Any Participant who is not a Part-Time Emp Participant shall be entitled to a Straight Life Pension that is calculated as set forth below.
- (1) Step One: Calculate the product of (a) the Participant's Final Average Compensation; multiplied by (b) the Participant's total number of years (and fractional years) of credited service determined pursuant to Section 3.2; multiplied by (c) 2.25%.
- (2) Step Two: Multiply one twelfth (1/12) by the lesser of (a) the amount calculated in Step One above; or (b) 80% of the Participant's Final Average Compensation. The amount calculated pursuant to the preceding sentence is the amount of the monthly benefit payable to the Participant as a Straight Life Pension.
- (b) <u>Part-Time Emp Participants.</u> Any Part-Time Emp Participant shall be entitled to a Straight Life Pension that is calculated as set forth below.
- (1) Step One: Calculate the product of (a) the Participant's Final Average Compensation; multiplied by (b) the Participant's total number of years (and fractional years) of credited service determined pursuant to Section 3.2; multiplied by (c) 2.00%.

(2) Step Two: Multiply one twelfth (1/12) by the amount calculated in Step One above. The amount calculated pursuant to the preceding sentence is the amount of the monthly benefit payable to the Participant as a Straight Life Pension.

If a Participant dies or becomes disabled (as described in Plan Section 5.6) while performing qualified military service (as defined in Code Section 414(u)), the Participant shall be considered for benefit accrual purposes to have resumed employment in accordance with the individual's reemployment rights under USERRA on the day preceding death or disability (as the case may be) and to have terminated employment on his actual date of death or disability.

- 5.2 <u>Reduced Early Retirement Benefit</u>. A Participant who is eligible for a reduced early retirement benefit hereunder shall be entitled to a reduced monthly Straight Life Pension equal to the participant's Accrued Benefit determined under the provisions of Section 5.1, reduced by a reduction factor. The reduction factor shall be 0.50%, multiplied by the number of months, rounded to the next higher number of months and not less than zero, by which the date of retirement precedes the date the member or vested former member would attain Normal Retirement Age. The amount of the reduction shall not be greater than 60%. Such amount shall further be adjusted as necessary based on the payment option(s) selected by the member or vested former member among those available
- 5.3 <u>Unreduced Early Retirement Benefit</u>. A Participant who is eligible for an unreduced early retirement benefit hereunder shall be entitled to an unreduced monthly Straight Life Pension equal to the participant's Accrued Benefit determined under the provisions of Section 5.1.
- 5.4 <u>Late Retirement Benefit</u>. A Participant may continue his or her employment with the Employer beyond his or her Normal Retirement Date. In such event, no retirement benefit will be paid to the Participant before the Participant's actual Retirement, subject, however, to any required minimum distributions pursuant to Section 5.9. Said late retirement benefit shall commence as set forth in Section 4.1 and shall be equal to the greater of (1) the Participant's Accrued Benefit determined as of his Late Retirement Date; or (2) the Actuarial Equivalent of the Participant's normal retirement benefit.

5.5 Death Benefit.

- (a) Surviving Spouse and Surviving Minor Child Benefits.
 - (1) Surviving Spouse of a Deceased Active Participant.
- [a] A retirement allowance shall be paid for life to the surviving spouse of a deceased active Participant if each of the following conditions is met:
- [i] The Participant was married to the surviving spouse at the time of the Participant's death.
- [ii] The Participant had not named another individual as monthly pension beneficiary at the time of death.
- [iii] The Participant was not receiving any form of benefits from the Plan at the time of death.

- [b] Payment of a retirement allowance to the surviving spouse of a deceased active Participant shall be effective as of the first day of the calendar month in which the Participant died. The amount of a surviving spouse's retirement allowance shall be 85% of the deceased active Participant's accrued retirement allowance (as determined pursuant to Section 5.1), but not less than the amount that would be paid under Section 5.5(c) if the surviving spouse had been designated as the monthly pension beneficiary by the Participant. The surviving spouse is required to complete and submit an application on the form prepared by the Plan Administrator, along with other forms and documents required, before benefits will commence.
- [c] The remarriage of a surviving spouse shall not affect the surviving spouse's eligibility to receive the retirement allowance.

(2) Surviving Spouse of Vested Former Participant

- [a] A retirement allowance shall be paid for life to the surviving spouse of a deceased vested former Participant if each of the following conditions is met:
- [i] The vested former Participant was married to the surviving spouse at the time of death.
- [ii] The vested former Participant had not named another individual as monthly pension beneficiary in the manner set forth in Section 5.5(c) at the time of death.
- [iii] The vested former Participant was not receiving any form of benefits from the Plan Administrator at the time of death.
- [b] A retirement allowance shall be paid to the surviving spouse of a deceased vested former Participant, the effective date of which shall be the first day of the month following the month of the later of:
- [i] the date on which the vested former Participant would have first satisfied the age and service requirements for retirement under Article IV; or
- [ii] the date on which a completed application on the form prepared by the Plan Administrator, along with other forms and documents required by the Plan Administrator, is received by the Plan Administrator.

Retroactive benefits shall not be payable for any period prior to the effective date as determined above.

The amount of a surviving spouse's retirement allowance shall be 85% of the deceased vested former Participant's accrued retirement allowance (as determined pursuant to subsection (4)), but not less than the amount that would be paid under Section 5.5(c) if the surviving spouse had been designated as the monthly pension beneficiary by the deceased vested former Participant.

[c] The remarriage of a surviving spouse shall not affect the surviving spouse's eligibility to receive the surviving spouse's retirement allowance.

- (3) Surviving Minor Child of a Deceased Active Participant or Vested Former Participant
- [a] A retirement allowance shall be paid to each surviving minor child of a deceased active Participant or a deceased vested former Participant if each of the following conditions is met:
- [i] The active Participant or vested former Participant does not have a Section 5.5(c) named monthly pension beneficiary other than the active Participant or vested former Participant's spouse at the time of death.
- [ii] The surviving minor child has not attained age 21 years as of the later of (i) the deceased active Participant or deceased vested former Participant's death, or (ii) the date on which a surviving spouse is no longer receiving a surviving spouse's retirement allowance.
- [b] Payment of a retirement allowance to a surviving minor child shall be effective on the first day of the month of the Participant's death, except that no retirement allowance shall be payable to any surviving minor child for any month for which the surviving spouse is paid a retirement allowance. An authorized representative of the minor child(ren) is required to complete and submit an application on the form prepared by the Plan Administrator, along with other forms and documents required, prior to commencement of benefits.

The amount of retirement allowance paid to each surviving minor child shall be an equal share (divided among the then-surviving minor children, if more than one) of 50% of the deceased active Participant or deceased vested former Participant's retirement allowance, as determined pursuant to subsection (4). Such amount shall be paid to each surviving minor child until the earlier of that surviving minor child's twenty-first birthday or the death of that child, with such amount then reallocated to remaining minor surviving children, if any, in equal shares.

- (4) <u>Calculation of Benefit</u>. For purposes of this Section 5.5(a), a deceased active Participant's or a deceased vested former Participant's accrued retirement benefit shall be computed under the following presumptions:
- [a] The deceased active Participant or deceased vested former Participant shall be presumed to have retired on the day preceding death.
- [b] The deceased active Participant or deceased vested former Participant shall be presumed to have elected the Straight Life Pension form or payment.
- (5) In the case of an active Participant who dies while performing qualified military service as defined in 414(u) of the Code, the survivors of a deceased active Participant who died on or after January 1, 2007 shall be entitled to any additional benefits, including benefit accruals relating to the period of qualified military service, provided under the Plan had the active Participant resumed and then terminated employment on account of death. Under section 414(u)(8)(C) of the Code, an active Participant who dies while performing qualified

military service shall be deemed to have made any employee contributions upon which benefit accruals described hereunder are contingent.

- Out of and In Course of Duty. If, after notice is provided to the Plan Administrator and appropriate review is made, the death of an active Participant is determined by the Plan Administrator to be the natural and proximate result, independent of all other causes, of a personal injury or disease arising out of and in the course of the active Participant's actual performance of duty with the Employer, the following additional provisions shall apply to Sections 5.5(a)(1) and 5.5(a)(3), all other provisions being the same.
- (1) The amount of retirement allowance paid to a surviving spouse shall not be less than 25% of the deceased active Participant's Final Average Compensation.
- (2) The amount of retirement allowance paid to surviving minor child(ren) shall not be less than equal shares of 25% of the deceased active Participant's final average compensation.

(c) Monthly Pension Beneficiary Retirement Allowance.

- (1) Monthly Pension Beneficiary Retirement Allowance on behalf of a Deceased Active Participant
- [a] An active Participant may name a monthly pension beneficiary for the exclusive purpose of being paid a monthly pension beneficiary retirement allowance under this section in the manner and subject to the conditions set out below.
- [i] The naming of a monthly pension beneficiary shall be in a format required by and submitted to the Plan Administrator, accompanied by such documentation and information as the Plan Administrator may require, and it must be received by the Plan Administrator prior to the death of the Participant.
- [ii] A spouse married to the Participant at the time of the designation of the monthly pension beneficiary election must consent (in a format required by and submitted to the Plan Administrator) to the naming of the monthly pension beneficiary if that monthly pension beneficiary is a person other than the spouse (unless otherwise provided by a judgment of divorce or other order of the court that meets the requirements of applicable law), unless the Plan Administrator determines that the spouse cannot be located or other extenuating circumstances.
- [iii] The designation of a monthly pension beneficiary may be revoked or changed in a format required by and submitted to the Plan Administrator by the Participant at any time prior to the earlier of the death of the Participant and the commencement of his or her retirement benefits, subject to all other requirements of the designation of a monthly pension beneficiary (including the consent of the Participant's current spouse, if any, in a format required by and submitted to the Plan Administrator, unless the Plan Administrator determines that the spouse cannot be located or other extenuating circumstances).

[iv] Any designation or status of a monthly pension beneficiary shall terminate immediately upon the marriage of the Participant, and that Participant's spouse shall then become the Participant's monthly pension beneficiary, unless otherwise required by a judgment of divorce or other order of the court that meets the requirements of applicable law. Nothing herein shall prevent a Participant from designating someone other than a current spouse as a monthly pension beneficiary, provided that such designation includes the consent of the Participant's current spouse, in a format required by and submitted to the Plan Administrator, unless the Plan Administrator determines that the spouse cannot be located or other extenuating circumstances.

[v] Notwithstanding anything in this Plan to the contrary, the designation or status of a spouse as a monthly pension beneficiary shall terminate immediately upon entry of a judgment of divorce terminating the marriage of the Participant and such spouse, unless otherwise required by a judgment of divorce or other order of the court that meets the requirements of applicable law. Nothing herein shall prevent a Participant from designating a former spouse as a monthly pension beneficiary subsequent to the entry of their judgment of divorce, provided that such designation is consistent with all other terms and requirements of this Plan (including the consent of the Participant's current spouse, if any, in a format required by and submitted to the Plan Administrator, unless the Plan Administrator determines that the spouse cannot be located or other extenuating circumstances).

[b] A monthly pension beneficiary retirement allowance shall be paid to the monthly pension beneficiary of a deceased active Participant for life, effective on the later of the first day of the month in which the deceased active Participant died, if both of the following conditions are met:

[i] The active Participant dies while still employed by

the Employer.

[ii] A spouse married to the active Participant at the time of death consented, in a format required by and submitted to the Plan Administrator, to the designation of the monthly pension beneficiary if it is someone other than the spouse married to the active Participant at the time of death (unless such designation is ordered by a judgment of divorce or other order of the court that meets the requirements of applicable law).

The monthly pension beneficiary is required to complete and submit to the Plan Administrator in a format required by the Plan Administrator, accompanied by such documentation and information as the Plan Administrator shall require prior to commencement of benefits.

[c] The amount of the monthly pension beneficiary retirement allowance payable to a monthly pension beneficiary of a deceased active Participant shall be computed under the following presumptions, notwithstanding the failure to satisfy the specific requirements of Article IV:

[i] The deceased active Participant shall be presumed to have retired on the later of (1) the day first day of the month preceding the Participant's death, or (2) the first day of the month following the month in which the monthly pension beneficiary completes and submits an application to the Plan Administrator, on forms provided by the Plan

Administrator and accompanied by such documentation and information as the Plan Administrator may require.

[ii] The deceased active Participant shall be presumed to have elected a joint and 100% survivor annuity and named the monthly pension beneficiary as monthly pension beneficiary.

[iii] If the monthly pension beneficiary is the spouse of the deceased active Participant, the amount of retirement allowance payable to the monthly pension beneficiary shall not be less than the amount that would have been payable under Section 5.5(a) if there had been no named monthly pension beneficiary.

(2) Monthly Pension Beneficiary Retirement Allowance on behalf of a Deceased Vested Former Participant

[a] A vested former Participant may name a monthly pension beneficiary for the exclusive purpose of being paid a monthly pension beneficiary retirement allowance under this section in the manner and subject to the conditions set out below.

[i] The naming of a monthly pension beneficiary shall be in a format required by and submitted to the Plan Administrator, accompanied by such documentation and information as the Plan Administrator may require, and it must be received by the Plan Administrator prior to the death of the Participant.

[ii] A spouse married to the Participant at the time of the designation of the monthly pension beneficiary election must consent (in a format required by and submitted to the Plan Administrator) to the naming of the monthly pension beneficiary if that monthly pension beneficiary other than the spouse (unless otherwise provided by a judgment of divorce or other order of the court that meets the requirements of applicable law), unless the Plan Administrator determines that the spouse cannot be located or other extenuating circumstances.

[iii] The designation of a monthly pension beneficiary may be revoked or changed (in a format required by and submitted to the Plan Administrator) by the Participant at any time prior to the earlier of the death of the Participant and the commencement of his or her retirement benefits, subject to all other requirements of the designation of a monthly pension beneficiary (including the consent of the Participant's current spouse, if any, in a format required by and submitted to the Plan Administrator, unless the Plan Administrator determines that the spouse cannot be located or other extenuating circumstances).

[iv] Any designation or status of a monthly pension beneficiary shall terminate immediately upon the marriage of the Participant, and that Participant's spouse shall then become the Participant's monthly pension beneficiary unless otherwise required by a judgment of divorce or other order of the court that meets the requirements of applicable law. Nothing herein shall prevent a Participant from designating someone other than a current spouse as a monthly pension beneficiary, provided that such designation includes the consent of the Participant's current spouse, in a format required by and submitted to the Plan Administrator, unless the Plan Administrator determines that the spouse cannot be located or other extenuating circumstances.

[v] Notwithstanding anything in this Plan to the contrary, the designation or status of a spouse as a monthly pension beneficiary shall terminate immediately upon entry of a judgment of divorce terminating the marriage of the Participant and such spouse, unless otherwise required by a judgment of divorce or other order of the court that meets the requirements of applicable law. Nothing herein shall prevent a Participant from designating a former spouse as a monthly pension beneficiary subsequent to the entry of their judgment of divorce, provided that such designation is consistent with all other terms and requirements of this Plan (including the consent of the Participant's current spouse, if any, in a format required by and submitted to the Plan Administrator, unless the Plan Administrator determines that the spouse cannot be located or other extenuating circumstances).

[b] A monthly pension beneficiary retirement allowance shall be paid to the monthly pension beneficiary of a deceased vested former Participant for life. Such allowance shall be effective on the latest of

[i] the first day of the month in which the vested former

Participant died, or

[ii] the first day of the month following the month in which the vested former Participant would have first satisfied age and service requirements for retirement, or

[iii] the first day of the month following the date that the monthly pension beneficiary completes and submits an application to the Plan Administrator on forms provided by the Plan Administrator and accompanied by such documentation and information as the Plan Administrator may require. Retroactive benefits shall not be payable.

[c] The amount of the monthly pension beneficiary retirement allowance payable to a monthly pension beneficiary shall be computed under the following presumptions, notwithstanding the failure to satisfy the specific requirements of Article IV:

[i] The deceased vested former Participant shall be presumed to have commenced receipt of the retirement allowance on the later of (1) the date the vested former Participant would have first satisfied age and service requirements for retirement, or (2) the date the monthly pension beneficiary completes and submits an application to the Plan Administrator in a format determined by the Plan Administrator, along with all other forms and documents required prior to commencement of payments.

[ii] The deceased vested former Participant shall be presumed to have elected a joint and 100% survivor annuity and named the designated monthly pension beneficiary as monthly pension beneficiary.

[iii] If the monthly pension beneficiary is the spouse of the deceased vested former Participant, the amount of retirement allowance payable to the monthly pension beneficiary shall not be less than the amount that would have been payable under Section 5.5(a) if there had been no named monthly pension beneficiary.

5.6 Disability Benefit.

- (a) An active Participant who becomes incapacitated such that the Participant is unable to continue his or her current employment with the Employer may apply for and receive a disability retirement benefit upon submission to the Plan Administrator of an application provided by the Plan Administrator, accompanied by such information the Plan Administrator may require, if each of the following conditions is met:
- (1) The Participant files an application for disability retirement benefits with the Plan Administrator within two (2) years after his or her employment with the Employer terminates as a result of such incapacity.
- (2) The Participant undergoes such medical examinations and tests as may be ordered by the Plan Administrator. Medical examination shall be made by or under the direction of a medical advisor selected by the Plan Administrator who shall issue a written medical examination report. "Medical examination" includes but is not limited to physical or psychiatric examination of the Participant, and/or a review of the Participant's application and medical records.
- (3) The Plan Administrator reviews the entire record, including the medical advisor's report, and concludes that the Participant is unable to continue his or her current employment with the Employer because, and as a result of his or her incapacity, and that the incapacity is likely to be permanent.
- (4) The Plan Administrator determines that the Employer is unable to continue to employ the Participant, and accommodate the Participant's incapacity with other work that is reasonably related to the Participant's past training, experience, education and compensation.
- (b) A disability retirement benefit shall be effective as of the first day of the calendar month next following the date on which the completed application for disability retirement benefits is received by the Plan Administrator, or, if later, the date of the Participant's termination of employment with the Employer resulting from the incapacity.
- (c) The amount of a disability retirement benefit shall be determined in accordance with the benefit programs that are applicable to the disability retiree's credited service.
- (d) If the Plan Administrator finds that the disability of an active Participant applying for disability retirement benefits was the natural and proximate result of a personal injury or disease arising out of and in the course of the Participant's actual performance of duty in the employ of the Employer, and not the aggravation of a preexisting injury or disease, then, if the Participant meets all other requirements of subsection (a), the disability retirement benefit determined under subsection (c) shall not be less than 25% of the Participant's Final Average Compensation. For purposes of this subsection (d), Final Average Compensation shall be determined for all credited service as of the date of termination of employment under the Final Average Compensation in effect on the termination date.
- (e) The Plan Administrator may require a disability retiree, prior to his or her attainment of Early Retirement Date, to undergo periodic medical examinations as described in Section 5.6(a)(2), but no more frequently than annually. If, upon medical examination and upon the advice of the medical advisor, the Plan Administrator determines that the disability retiree is

no longer incapacitated from performing his or her former job with the Employer, disability retirement benefits shall be terminated upon 60 days' prior notice. If a disability retiree fails or refuses to submit to or cooperate with a medical examination, disability retirement benefits shall be immediately suspended with notice to the disability retiree. If a disability retiree fails or refuses to submit to or cooperate with a medical examination for six months after such suspension, his or her disability retirement benefits shall be terminated. A disability retiree whose benefits have been discontinued shall be deemed a former disability retiree.

- (f) If a former disability retiree returns to covered employment with the Employer within five years after the date of discontinuation of disability retirement benefits, the former disability retiree shall recommence participation in the Plan, with all credited service the individual had accrued at the time of disability retirement.
- (g) A former disability retiree who does not again become a Participant described in subsection (f) shall become a vested former Participant.
- (h) Service shall not be credited for the period during which a disability retiree is being paid or was paid a disability retirement allowance.
- 5.7 <u>Deferred Vested Pension Benefit</u>. A Participant who is eligible for a deferred vested pension benefit shall be entitled to a retirement benefit that is Actuarially Equivalent to the Participant's Accrued Benefit determined in accordance with Section 5.1. Effective as of the Effective Date, each Participant shall be fully vested in his or her Accrued Benefit.

5.8 Payment of Benefit.

- Participant is married at the retirement allowance effective date, benefits shall be paid in the form of a joint and 100% survivor annuity, with the spouse as monthly pension beneficiary, unless the retiring Participant or vested former Participant elects another form of payment, and his or her spouse consents to that election in a signed writing. This requirement may be waived if it is determined by the Plan Administrator that the consent of the spouse cannot be obtained because of extenuating circumstances. The remarriage of a surviving spouse shall have no effect on the right of the surviving spouse to receive a survivor benefit. If a retiring Participant or vested former Participant is not married at the retirement allowance effective date, benefits shall be paid in the form of a Straight Life Pension, unless the Participant elects in writing to receive his or her benefit in one of the forms listed in Section 5.8(b).
- (b) Optional Benefits. In lieu of the benefits provided for in Section 5.8(a), a Participant may elect an Actuarially Equivalent optional benefit, provided that any spousal consent required by Section 5.8(a) is provided. Such Actuarially Equivalent optional benefit may be:
- (1) A monthly benefit payable for the life of the Participant with a percentage (50%, 75%, or 100%) of such monthly benefit as specified by the Participant continued to his or her Eligible Spouse or other designated beneficiary after his or her death. If the named monthly beneficiary dies before the Participant, effective as of the date of the Plan Administrator's receipt of such documentation as the Plan Administrator shall require, the Participant's benefit shall be converted to a straight life annuity, and payable thereafter for the Participant's remaining life. If the required documentation is received by the Plan Administrator within six months after the

beneficiary's death, then the Participant's benefit shall be converted to a straight life annuity retroactive to the month following the month of the beneficiary's death. If the required documentation is received by MERS later than six months after the beneficiary's death, then the retiree's benefit shall be converted to a straight life annuity effective as of the month following the month in which the required documentation is received by the Plan Administrator, with no retroactive payment.

(2) A monthly benefit payable for the life of the Participant with a guaranteed period of 60 months, 120 months, 180 months, or 240 months, as specified by the Participant. Under this form of benefit payment, the retiree is paid an actuarially reduced retirement allowance for life. If the retiree dies prior to the expiration of the guaranteed period selected by the retiree, the named monthly pension beneficiary is paid a survivor benefit equal to the full amount of the reduced retirement allowance for the balance of the guaranteed period remaining after the death of the retiree, after which benefits to the named monthly pension beneficiary end. Immediately upon the death of the monthly pension beneficiary before the expiration of the guaranteed period, or at any other time, the retiree may replace the current monthly pension beneficiary with a new monthly pension beneficiary by completing and submitting a written designation form provided by the Plan Administrator. If the retiree's election of the this form of payment required that his or her spouse consent to that election in a signed writing, and the retiree remains married to that spouse, written consent by that spouse to retiree's designation of the new monthly pension beneficiary is required; this requirement may be waived if it is determined by the Plan Administrator that the signature of the participant's spouse cannot be obtained because of extenuating circumstances. If the monthly pension beneficiary dies before the expiration of the guaranteed period (and after the death of the retiree), the single sum actuarial present value of any remaining guaranteed retirement allowance payments shall be paid to the estate of such monthly pension beneficiary.

The aggregate of the benefit payments that are expected to be paid to the Participant and his designated beneficiary under any of the above optional forms of payment shall be the Actuarial Equivalent of the straight life annuity that is described in subsection (a) above.

Divorce. If a retiree receiving benefits in the form of a joint and survivor annuity under Section 5.8(b)(1) is divorced from the spouse who was his or her named beneficiary at the time benefit payments commenced, the designation of the former spouse as monthly pension beneficiary shall terminate immediately upon entry of the judgment of divorce and the retiree shall be entitled to receive his or her retirement allowance in the form of a Straight Life Pension (no new beneficiary can be named), effective on and after the first day of the month next following the receipt by the Plan Administrator of a true copy of the judgment of divorce and/or other order of the court, together with any other documentation required by the Plan Administrator, unless the Plan Administrator receives a copy of the judgment of divorce and/or other order of the court that maintains the survivor beneficiary rights of such ex-spouse. If a retiree receiving benefits in the form a life annuity with guaranteed period under Section 5.8(b)(2) is divorced from the spouse who was his/her named beneficiary at the time benefit payments commenced, the designation of the former spouse as the retiree's beneficiary shall terminate immediately upon entry of the judgment of divorce, unless the judgment of divorce and/or other order of the court maintains the beneficiary rights of such ex-spouse. Where the ex-spouse's rights are terminated, and the retiree's period certain has not yet expired, the retiree may name a new beneficiary (or beneficiaries) for

the remaining period certain by submitting instructions to the Plan Administrator in a format determined by the Plan Administrator based on procedures established by the Plan Administrator.

- (d) <u>Eligible Domestic Relations Orders</u>. Certain terms and requirements for payment of retirement allowances in forms provided in this section are subject to eligible domestic relations orders under the Eligible Domestic Relations Order Act.
- (e) <u>Written Explanation.</u> Each Participant or vested former Participant who applies for a retirement allowance shall be given a written explanation, prior to the effective date of retirement, of the optional forms of payment provided in this section.

In the event of distribution of benefits pursuant to Section 401(a)(9) of the Code where the Participant or the vested former Participant fails to make an election of benefit form pursuant to this Section, benefits to an unmarried Participant or vested former Participant shall be paid in the a Single Life Annuity, and benefits to a married Participant or vested former Participant shall be paid in the form of a joint and 100% survivor annuity.

- 5.9 <u>Maximum Annual Pension Benefit by Law</u>. Regardless of any other provision in this Plan to the contrary, in no event shall the maximum annual pension benefit payable to any Participant from this Plan exceed the limits set forth below.
- (a) The limitations of this Section 5.9 shall apply in Limitation Years beginning on or after July 1, 2007, except as otherwise provided herein.
- (b) The Annual Benefit otherwise payable to a Participant under the Plan at any time will not exceed the Maximum Permissible Benefit. If the benefit the Participant would otherwise accrue in a Limitation Year would produce an Annual Benefit in excess of the Maximum Permissible Benefit, the benefit shall be limited (or the rate of accrual will be reduced) to the extent necessary so that the benefit does not exceed the Maximum Permissible Benefit.
- (c) If a Participant is, or has ever been, a participant in another qualified defined benefit plan (without regard to whether the plan has been terminated) maintained by the Employer or a Predecessor Employer, the sum of the Participant's Annual Benefits from all such Plans may not exceed the Maximum Permissible Benefit. Where the Participant's Employer-provided benefits under all such defined benefit plans (determined as of the same age) would exceed the Maximum Permissible Benefit applicable at that age, the Participant's benefit accrual will be limited in accordance with this Section 5.9.
- (d) The application of the limitations of this Section shall not cause the Maximum Permissible Benefit for any Participant to be less than the Participant's Accrued Benefit under all the defined benefit plans of the Employer or a Predecessor Employer as of the end of the last Limitation Year beginning before July 1, 2007 under provisions of the plans that were both adopted and in effect before April 5, 2007. The preceding sentence applies only if the provisions of such defined benefit plans that were both adopted and in effect before April 5, 2007 satisfied the applicable requirements of statutory provisions, regulations and other published guidance relating to Code Section 415 in effect as of the end of the last Limitation Year beginning before July 1, 2007, as described in Regulation Section 1.415(a)-1(g)(4). The preceding sentence applies only if all such defined benefit plans met the requirements of Code Section 415, for all Limitation Years beginning before May 6, 1986. The preceding sentence applies only if such defined benefit

plans met the requirements of Code Section 415, for all Limitation Years beginning before January 1, 1987.

(e) The limitations of this Section 5.9 will be determined and applied taking into account the rules in Section 5.9(g) below.

(f) Definitions.

(1) <u>Annual Benefit</u>. A benefit which is payable annually in the form of a straight life annuity. Except as provided below, where a benefit is payable in a form other than a straight life annuity, the benefit must be adjusted to an actuarially equivalent straight life annuity that begins at the same time as such other form of benefit and is payable on the first day of each month, before applying the limitations of this Section. For a Participant who has or will have distributions commencing at more than one annuity starting date, the Annual Benefit shall be determined as of each such annuity starting date (and shall satisfy the limitations of this Section as of each such date), actuarially adjusting for past and future distributions of benefits commencing at the other annuity starting dates. For this purpose, the determination of whether a new starting date has occurred shall be made without regard to Regulation Section 1.401(a)-20, Q&A 10(d), and with regard to Regulation Section 1.415(b)-1(b)(1)(iii)(B) and (C).

No actuarial adjustment to the benefit shall be made for (a) survivor benefits payable to a surviving spouse under a qualified joint and survivor annuity to the extent such benefits would not be payable if the Participant's benefit were paid in another form; (b) benefits that are not directly related to retirement benefits (such as a qualified disability benefit, preretirement incidental death benefits, and post-retirement medical benefits); or (c) the inclusion in the form of benefit of an automatic benefit increase feature, provided the form of benefit is not subject to Code Section 417(e)(3) and would otherwise satisfy the limitations of this Section, and the plan provides that the amount payable under the form of benefit in any Limitation Year shall not exceed the limits of this Section applicable at the annuity starting date, as increased in subsequent years pursuant to Code Section 415(d). For this purpose, an automatic benefit increase feature is included in a form of benefit if the form of benefit provides for automatic, periodic increases to the benefits paid in that form.

The determination of the Annual Benefit shall take into account Social Security supplements described in Code Section 411(a)(9) and benefits transferred from another defined benefit plan, other than transfers of distributable benefits pursuant to Regulation Section 1.411(d)-4, Q&A-3(c), but shall disregard benefits attributable to employee contributions or rollover contributions.

Effective for distributions in Plan Years beginning after December 31, 2003, the determination of actuarial equivalence of forms of benefit other than a straight life annuity shall be made in accordance with Section 5.9(f)(1)[a] or [b] below.

[a] Benefit Forms Not Subject to Code Section 417(e)(3). The straight life annuity that is actuarially equivalent to the Participant's form of benefit shall be determined under this Section 5.9(f)(1)[a] if the form of the Participant's benefit is either (i) a nondecreasing annuity (other than a straight life annuity) payable for a period of not less than the life of the Participant (or, in the case of a qualified pre-retirement survivor annuity, the life of the surviving spouse), or (ii) an annuity that decreases during the life of the Participant merely because

of (a) the death of the survivor annuitant (but only if the reduction is not below 50% of the benefit payable before the death of the survivor annuitant), or (b) the cessation or reduction of Social Security supplements or qualified disability payments (as defined in Code Section 401(a)(11)).

[i] <u>Limitation Years beginning before July 1, 2007</u>. For Limitation Years beginning before July 1, 2007, the actuarially equivalent straight life annuity is equal to the annual amount of the straight life annuity commencing at the same annuity starting date that has the same actuarial present value as the Participant's form of benefit computed using whichever of the following produces the greater annual amount: (I) the interest rate specified in Section 2.4(a) of the Plan and the mortality table (or other tabular factor) specified in Section 2.4(a) of the Plan for adjusting benefits in the same form; and (II) a 5% interest rate assumption and the applicable mortality table defined in Section 2.4(b) of the Plan for that annuity starting date.

[ii] <u>Limitation Years beginning on or after July 1, 2007</u>. For Limitation Years beginning on or after July 1, 2007, the actuarially equivalent straight life annuity is equal to the greater of (a) the annual amount of the straight life annuity (if any) payable to the Participant under the Plan commencing at the same annuity starting date as the Participant's form of benefit; and (b) the annual amount of the straight life annuity commencing at the same annuity starting date that has the same actuarial present value as the Participant's form of benefit, computed using a 5% interest rate assumption and the applicable mortality table defined in Section 2.4(b) of the Plan for that annuity starting date.

[b] <u>Benefit Forms Subject to Code Section 417(e)(3)</u>. The straight life annuity that is actuarially equivalent to the Participant's form of benefit shall be determined under this paragraph if the form of the Participant's benefit is other than a benefit form described in Section 5.9(f)(1)[a]. In this case, the actuarially equivalent straight life annuity shall be determined as follows:

[i] Annuity Starting Date in Plan Years Beginning After 2005. If the annuity starting date of the Participant's benefit occurs during a Plan Year beginning after 2005, the actuarially equivalent straight life annuity is equal to the greatest of [a] the annual amount of the straight life annuity commencing at the same annuity starting date that has the same actuarial present value as the Participant's form of benefit, computed using the interest rate specified in Section 2.2(a) of the Plan and the mortality table (or other tabular factor) specified in Section 2.2(a) of the Plan for adjusting benefits in the same form; [b] the annual amount of the straight life annuity commencing at the same annuity starting date that has the same actuarial present value as the Participant's form of benefit, computed using a 5.5% interest rate assumption and the applicable mortality table defined in Section 2.2(b) of the Plan; and [c] the annual amount of the straight life annuity commencing at the same annuity starting date that has the same actuarial present value as the Participant's form of benefit, computed using the applicable interest rate defined in Section 2.2(b) of the Plan and the applicable mortality table defined in Section 2.2(b) of the Plan, divided by 1.05.

However, effective for benefits with annuity starting dates during Limitation Years beginning after December 31, 2008, item [b][i][c] in the foregoing paragraph does not apply to a plan maintained by an eligible employer under Code Section 408(p)(2)(C)(i) (generally, an employer that had no more than 100 employees who received at least \$5,000 of compensation from the employer during the preceding year).

[ii] Annuity Starting Date in Plan Years Beginning in 2004 or 2005. If the annuity starting date of the Participant's form of benefit is in a Plan Year beginning in 2004 or 2005, the actuarially equivalent straight life annuity is equal to the annual amount of the straight life annuity commencing at the same annuity starting date that has the same actuarial present value as the Participant's form of benefit, computed using whichever of the following produces the greater annual amount: [a] the interest rate specified in Section 2.2(a) of the Plan and the mortality table (or other tabular factor) specified in Section 2.2(a) of the Plan for adjusting benefits in the same form; and [b] a 5.5% interest rate assumption and the applicable mortality table defined in Section 2.2(b) of the Plan.

If the annuity starting date of the Participant's benefit is on or after the first day of the first Plan Year beginning in 2004 and before December 31, 2004, the application of this Section 5.9(f)(1)[b][ii] shall not cause the amount payable under the Participant's form of benefit to be less than the benefit calculated under the Plan, taking into account the limitations of this Section, except that the actuarially equivalent straight life annuity is equal to the annual amount of the straight life annuity commencing at the same annuity starting date that has the same actuarial present value as the Participant's form of benefit, computed using whichever of the following produces the greatest annual amount:

- a. the interest rate specified in Section 2.2(a) of the Plan and the mortality table (or other tabular factor) specified in Section 2.2(a) of the Plan for adjusting benefits in the same form (as provided under the Plan as of the date of distribution);
- b. the applicable interest rate defined in Section 2.2(b) of the Plan and the applicable mortality table defined in Section 2.2(b) of the Plan (as provided under the Plan as of the date of distribution); and
- c. the applicable interest rate defined in Section 2.2(b) of the Plan (as in effect on the last day of the Plan Year beginning before January 1, 2004, under provisions of the Plan then adopted and in effect) and the applicable mortality table defined in Section 2.2(b) of the Plan.
- (2) <u>Compensation</u>. wages as defined in Code Section 3401(a) and all other payments of compensation by the Employer (in the course of the Employer's trade or business) for a Plan Year for which the Employer is required to furnish the Participant a written statement under Code Sections 6041(d), 6051(a)(3) and 6052. Compensation must be determined without regard to any rules under Code Section 3401(a) that limit the remuneration included in wages based on the nature or location of the employment or the services performed (such as the exception for agricultural labor in Code Section 3401(a)(2)). Notwithstanding the foregoing, compensation purposes of this Section 5.9 shall include any elective deferral (as defined in Code Section 402(g)(3)), and any amount which is contributed by the Employer at the election of the Participant pursuant to a salary reduction agreement and which is not includible in the gross income of the Participant by reason of Code Sections 125, 132(f)(4), 402(e)(3), 402(h)(1)(B), 403(b) or 457(b), and Employee contributions described in Code Section 414(h)(2) that are treated as Employer contributions.

For Limitation Years beginning on or after July 1, 2007, Compensation for a Limitation Year shall also include compensation paid by the later of 2½ months after an Employee's severance from employment with the Employer maintaining the Plan

or the end of the Limitation Year that includes the date of the Employee's severance from employment with the Employer maintaining the Plan if:

[a] the payment is regular compensation for services during the Employee's regular working hours, or compensation for services outside the Employee's regular working hours (such as overtime or shift differential), commissions, bonuses, or other similar payments, and, absent a severance from employment, the payments would have been paid to the Employee while the Employee continued in employment with the Employer;

[b] the payment is for unused accrued bona fide sick, vacation or other leave that the Employee would have been able to use if employment had continued; or

[c] the payment is received by the Employee pursuant to a nonqualified unfunded deferred compensation plan and would have been paid at the same time if employment had continued, but only to the extent includible in gross income.

Any payments not described above shall not be considered Compensation if paid after severance from employment, even if they are paid by the later of 2½ months after the date of severance from employment or the end of the Limitation Year that includes the date of severance from employment.

Compensation for purposes of this Section 5.9 shall not include payments to an individual who does not currently perform services for the Employer by reason of qualified military service (within the meaning of Code Section 414(u)(1)) to the extent these payments do not exceed the amounts the individual would have received if the individual had continued to perform services for the Employer rather than entering qualified military service and salary continuation payments to a Participant who is permanently and totally disabled, as defined in Code Section 22(e)(3).

Effective for Plan Years beginning after December 31, 2008, an individual who is receiving a differential wage payment, as defined by Code Section 3401(h)(2), shall be treated as an Employee of the Employer making the payment. The differential wage payment shall be treated as compensation for purposes of Code Section 415(c)(3) and Section 1.415(c)-2 of the Regulations and the Plan shall not be treated as failing to meet the requirements of any provisions described in Code Section 414(u)(1)(C) by reason of any contribution or benefit that is based on the differential wage payment.

Back pay, within the meaning of Regulation Section 1.415(c)-2(g)(8), shall be treated as Compensation for the Limitation Year to which the back pay relates to the extent the back pay represents wages and compensation that would otherwise be included under this definition.

Except as otherwise provided herein, if, in connection with the adoption of any amendment, the definition compensation for purposes of this Section 5.9 has been modified, then for Plan Years prior to the Plan Year that includes the adoption date of such amendment, compensation for purposes of this Section 5.9 means compensation determined pursuant to the terms of the Plan then in effect.

Compensation shall not include amounts paid as compensation to a non-resident alien, as defined in Code Section 7701(b)(1)(B), who is not a Participant in the Plan to the extent the compensation is excludable from gross income and is not effectively connected with the conduct of a trade or business within the United States.

- ending after December 31, 2001, the Defined Benefit Dollar Limitation is \$160,000, automatically adjusted under Code Section 415(d), effective January 1 of each year, as published in the Internal Revenue Bulletin, and payable in the form of a straight life annuity. The new limitation shall apply to Limitation Years ending with or within the calendar year of the date of the adjustment, but a Participant's benefits shall not reflect the adjusted limit prior to January 1 of that calendar year. The automatic annual adjustment of the Defined Benefit Dollar Limitation under Code Section 415(d) shall apply to Participants who have had a separation from employment.
- (4) <u>Employer</u>: For purposes of this Section 5.9, Employer shall mean the employer that adopts this Plan, and all members of a controlled group of corporations (as defined in Code Section 414(b), as modified by Code Section 415(h)), all commonly controlled trades or businesses (as defined in Code Section 414(c) as modified, except in the case of a brother-sister group of trades or businesses under common control, by Code Section 415(h)), or affiliated service groups (as defined in Code Section 414(m)) of which the Employer is a part, and any other entity required to be aggregated with the Employer pursuant to Regulations under Code Section 414(o).
- (5) <u>Formerly Affiliated Plan of the Employer</u>. A plan that, immediately prior to the cessation of affiliation, was actually maintained by the Employer and, immediately after the cessation of affiliation, is not actually maintained by the Employer. For this purpose, cessation of affiliation means the event that causes an entity to no longer be considered the Employer, such as the sale of a member of a controlled group of corporations, as defined in Code Section 414(b), as modified by Code Section 415(h), to an unrelated corporation, or that causes a plan to not actually be maintained by the Employer, such as transfer of plan sponsorship outside a controlled group.
- (6) <u>Limitation Year.</u> A calendar year, or the 12- consecutive-month period elected by the Employer. All qualified Plans maintained by the Employer must use the same Limitation Year. The Limitation Year for purposes of this Plan shall be the same as the Plan Year. If the Limitation Year is amended to a different 12-consecutive-month period, the new Limitation Year must begin on a date within the Limitation Year in which the amendment is made.
- (7) <u>Maximum Permissible Benefit</u>. The Defined Benefit Dollar Limitation (adjusted where required, as provided below).
- [a] Adjustment for Less Than 10 Years of Participation or Service. If the Participant has less than 10 Years of Participation in the Plan, the Defined Benefit Dollar Limitation shall be multiplied by a fraction [i] the numerator of which is the number of Years (or part thereof, but not less than one year) of Participation in the Plan, and [ii] the denominator of which is 10.
- [b] <u>Adjustment of Defined Benefit Dollar Limitation for Benefit Commencement Before Age 62 or after Age 65</u>. Effective for benefits commencing in Limitation Years ending after December 31, 2001, the Defined Benefit Dollar Limitation shall be adjusted if the annuity starting date of the Participant's benefit is before age 62 or after age 65. If the annuity

starting date is before age 62, the Defined Benefit Dollar Limitation shall be adjusted under Section 5.9(f)(7)[b][i], as modified by Section 5.9(f)(7)[b][iii]. If the annuity starting date is after age 65, the Defined Benefit Dollar Limitation shall be adjusted under Section 5.9(f)(7)[b][ii], as modified by Section 5.9(f)(7)[b][iii].

[i] Adjustment of Defined Benefit Dollar Limitation for Benefit Commencement Before Age 62.

a. <u>Limitation Years Beginning Before July 1, 2007</u>. If the annuity starting date for the Participant's benefit is prior to age 62 and occurs in a Limitation Year beginning before July 1, 2007, the Defined Benefit Dollar Limitation for the Participant's annuity starting date is the annual amount of a benefit payable in the form of a straight life annuity commencing at the Participant's annuity starting date that is the actuarial equivalent of the Defined Benefit Dollar Limitation (adjusted under Section 5.9(f)(7)[a] for Years of Participation less than 10, if required) with actuarial equivalence computed using whichever of the following produces the smaller annual amount: (1) the interest rate specified in Section 2.2(a) of the Plan and the mortality table (or other tabular factor) specified in Section 2.4(a) of the Plan; or (2) a 5% interest rate assumption and the applicable mortality table as defined in Section 2.2(b) of the Plan. This reduction shall not apply to certain disability and survivor benefits that are described in Internal Revenue Code Section 415(b)(2)(I).

b. <u>Limitation Years Beginning on or After</u>

<u>July 1, 2007</u>.

Commencing Straight Life Annuity Payable at Both Age 62 and the Age of Benefit Commencement. If the annuity starting date for the Participant's benefit is prior to age 62 and occurs in a Limitation Year beginning on or after July 1, 2007, and the Plan does not have an immediately commencing straight life annuity payable at both age 62 and the age of benefit commencement, the Defined Benefit Dollar Limitation for the Participant's annuity starting date is the annual amount of a benefit payable in the form of a straight life annuity commencing at the Participant's annuity starting date that is the actuarial equivalent of the Defined Benefit Dollar Limitation (adjusted under Section 5.9(f)(7)[a] for Years of Participation less than 10, if required) with actuarial equivalence computed using a 5% interest rate assumption and the applicable mortality table for the annuity starting date as defined in Section 2.2(b) of the Plan (and expressing the Participant's age based on completed calendar months as of the annuity starting date). This reduction shall not apply to certain disability and survivor benefits that are described in Internal Revenue Code Section 415(b)(2)(I).

Straight Life Annuity Payable at Both Age 62 and the Age of Benefit Commencement. If the annuity starting date for the Participant's benefit is prior to age 62 and occurs in a Limitation Year beginning on or after July 1, 2007, and the Plan has an immediately commencing straight life annuity payable at both age 62 and the age of benefit commencement, the Defined Benefit Dollar Limitation for the Participant's annuity starting date is the lesser of the limitation determined under Section 5.9(f)(7)[b][i]a. and the Defined Benefit Dollar Limitation (adjusted under Section 5.9(f)(7)[a] for Years of Participation less than 10, if required) multiplied by the ratio of the annual amount of the immediately commencing straight life annuity under the Plan at the Participant's

annuity starting date to the annual amount of the immediately commencing straight life annuity under the Plan at age 62, both determined without applying the limitations of this Article.

c. Notwithstanding any other provisions of this Section 5.9(f)(7)[b][i], the age-adjusted dollar limit that is applicable to a Participant shall not decrease on account of an increase in age or performance of additional service.

[ii] Adjustment of Defined Benefit Dollar Limitation for Benefit Commencement After Age 65.

a. <u>Limitation Years Beginning Before July 1, 2007</u>. If the annuity starting date for the Participant's benefit is after age 65 and occurs in a Limitation Year beginning before July 1, 2007, the Defined Benefit Dollar Limitation for the Participant's annuity starting date is the annual amount of a benefit payable in the form of a straight life annuity commencing at the Participant's annuity starting date that is the actuarial equivalent of the Defined Benefit Dollar Limitation (adjusted under Section 5.9(f)(7)[a] for Years of Participation less than 10, if required) with actuarial equivalence computed using whichever of the following produces the smaller annual amount: (1) the interest rate specified in Section 2.2(a) of the Plan and the mortality table (or other tabular factor) specified in Section 2.2(a) of the Plan; or (2) a 5% interest rate, assumption and the applicable mortality table as defined in Section 2.2(b) of the Plan.

b. <u>Limitation Years Beginning on or After</u>

July 1, 2007.

Commencing Straight Life Annuity Payable at Both Age 65 and the Age of Benefit Commencement. If the annuity starting date for the Participant's benefit is after age 65 and occurs in a Limitation Year beginning on or after July 1, 2007, and the Plan does not have an immediately commencing straight life annuity payable at both age 65 and the age of benefit commencement, the Defined Benefit Dollar Limitation at the Participant's annuity starting date is the annual amount of a benefit payable in the form of a straight life annuity commencing at the Participant's annuity starting date that is the actuarial equivalent of the Defined Benefit Dollar Limitation (adjusted under Section 5.9(f)(7)[a] for Years of Participation less than 10, if required), with actuarial equivalence computed using a 5% interest rate assumption and the applicable mortality table for that annuity starting date as defined in Section 2.2(b) of the Plan (and expressing the Participant's age based on completed calendar months as of the annuity starting date).

Straight Life Annuity Payable at Both Age 65 and the Age of Benefit Commencement. If the annuity starting date for the Participant's benefit is after age 65 and occurs in a Limitation Year beginning on or after July 1, 2007, and the Plan has an immediately commencing straight life annuity payable at both age 65 and the age of benefit commencement, the Defined Benefit Dollar Limitation at the Participant's annuity starting date is the lesser of the limitation determined under Section 5.9(f)(7)[b][ii]a. and the Defined Benefit Dollar Limitation (adjusted under Section 5.9(f)(7)[a] for Years of Participation less than 10, if required) multiplied by the ratio of the annual amount of the adjusted immediately commencing straight life annuity under the Plan at the Participant's annuity starting date to the annual amount of the adjusted immediately commencing straight life annuity under the Plan at age 65, both determined without applying the limitations of

this Section. For this purpose, the adjusted immediately commencing straight life annuity under the Plan at the Participant's annuity starting date is the annual amount of such annuity payable to the Participant, computed disregarding the Participant's accruals after age 65 but including actuarial adjustments even if those actuarial adjustments are used to offset accruals; and the adjusted immediately commencing straight life annuity under the Plan at age 65 is the annual amount of such annuity that would be payable under the Plan to a hypothetical Participant who is age 65 and has the same accrued benefit as the Participant.

[iii] Notwithstanding the other requirements of this Section 5.9(f)(7)[b], for purposes of adjusting the Defined Benefit Dollar Limitation for the Participant's annuity starting date under Section 5.9(f)(7)[b][i]a., 5.9(f)(7)[b][i]b.1., 5.9(f)(7)[b][ii]a. or 5.9(f)(7)[b][ii]b.1., no adjustment shall be made to reflect the probability of a Participant's death between the annuity starting date and age 62, or between age 65 and the annuity starting date, as applicable, if benefits are not forfeited upon the death of the Participant prior to the annuity starting date. To the extent benefits are forfeited upon death before the annuity starting date, such an adjustment shall be made. For this purpose, no forfeiture shall be treated as occurring upon the Participant's death if the Plan does not charge Participants for providing a qualified preretirement survivor annuity, as defined in Code Section 417(c) upon the Participant's death.

- [c] <u>Minimum Benefit Permitted</u>. Notwithstanding anything else in this Section to the contrary, the benefit otherwise accrued or payable to a Participant under this Plan shall be deemed not to exceed the Maximum Permissible Benefit if:
- [i] the retirement benefits payable for a Limitation Year under any form of benefit with respect to such Participant under this Plan and under all other defined benefit plans (without regard to whether a Plan has been terminated) ever maintained by the Employer do not exceed \$10,000 multiplied by a fraction—[a] the numerator of which is the Participant's number of Years (or part thereof, but not less than one year) of Service (not to exceed 10) with the Employer, and [b] the denominator of which is 10; and
- [ii] the Employer (or a Predecessor Employer) has not at any time maintained a defined contribution plan in which the Participant participated (for this purpose, mandatory employee contributions under a defined benefit plan, individual medical accounts under Code Section 401(h), and accounts for postretirement medical benefits established under Code Section 419A(d)(1) are not considered a separate defined contribution plan).
- (8) <u>Predecessor Employer</u>. If the Employer maintains a plan that provides a benefit which the Participant accrued while performing services for a former employer, the former employer is a Predecessor Employer with respect to the Participant in the Plan. A former entity that antedates the Employer is also a Predecessor Employer with respect to a Participant if, under the facts and circumstances, the Employer constitutes a continuation of all or a portion of the trade or business of the former entity.
- (9) <u>Severance from Employment</u>. An Employee has a severance from employment when the Employee ceases to be an Employee of the Employer maintaining the Plan. An Employee does not have a severance from employment if, in connection with a change of employment, the Employee's new employer maintains the Plan with respect to the Employee.

Year of Participation. The Participant shall be credited with a Year (10)of Participation (computed to fractional parts of a year) for each accrual computation period for which the following conditions are met: [a] the Participant is credited with at least the number of Hours of Service (or period of service if the elapsed time method is used) for benefit accrual purposes, required under the terms of the Plan in order to accrue a benefit for the accrual computation period, and [b] the Participant is included as a Participant under the eligibility provisions of the Plan for at least one day of the accrual computation period. If these two conditions are met, the portion of a Year of Participation credited to the Participant shall equal the amount of benefit accrual service credited to the Participant for such accrual computation period. Participant who is permanently and totally disabled within the meaning of Code Section 415(c)(3)(C)(i) for an accrual computation period shall receive a Year of Participation with respect to that period. In addition, for a Participant to receive a Year of Participation (or part thereof) for an accrual computation period, the Plan must be established no later than the last day of such accrual computation period. In no event shall more than one Year of Participation be credited for any 12month period.

(g) Other Rules.

- (1) <u>Benefits Under Terminated Plans</u>. If a defined benefit plan maintained by the Employer has terminated with sufficient assets for the payment of benefit liabilities of all Plan Participants and a Participant in the Plan has not yet commenced benefits under the Plan, the benefits provided pursuant to the annuities purchased to provide the Participant's benefits under the terminated Plan at each possible annuity starting date shall be taken into account in applying the limitations of this Section. If there are not sufficient assets for the payment of all Participants' benefit liabilities, the benefits taken into account shall be the benefits that are actually provided to the Participant under the terminated Plan.
- (2) Benefits Transferred From the Plan. If a Participant's benefits under a defined benefit plan maintained by the Employer are transferred to another defined benefit plan maintained by the Employer and the transfer is not a transfer of distributable benefits pursuant to Regulation Section 1.411(d)-4, Q&A-3(c), the transferred benefits are not treated as being provided under the transferor plan (but are taken into account as benefits provided under the transferee plan). If a Participant's benefits under a defined benefit plan maintained by the Employer are transferred to another defined benefit plan that is not maintained by the Employer and the transfer is not a transfer of distributable benefits pursuant to Regulation Section 1.411(d)-4, Q&A-3(c), the transferred benefits are treated by the Employer's plan as if such benefits were provided under annuities purchased to provide benefits under a plan maintained by the Employer that terminated immediately prior to the transfer with sufficient assets to pay all Participants' benefit liabilities under the Plan. If a Participant's benefits under a defined benefit plan maintained by the Employer are transferred to another defined benefit plan in a transfer of distributable benefits pursuant to Regulation Section 1.411(d)-4, Q&A-3(c), the amount transferred is treated as a benefit paid from the transferor plan.
- (3) <u>Formerly Affiliated Plans of the Employer</u>. A formerly affiliated plan of the Employer shall be treated as a plan maintained by the Employer, but the formerly affiliated plan shall be treated as if it had terminated immediately prior to the cessation of affiliation with sufficient assets to pay Participants' benefit liabilities under the Plan and had purchased annuities to provide benefits.

- (4) <u>Plans of a Predecessor Employer</u>. If the Employer maintains a defined benefit plan that provides benefits accrued by a Participant while performing services for a Predecessor Employer, the Participant's benefits under a plan maintained by the Predecessor Employer shall be treated as provided under a plan maintained by the Employer. However, for this purpose, the plan of the Predecessor Employer shall be treated as if it had terminated immediately prior to the event giving rise to the Predecessor Employer relationship with sufficient assets to pay Participants' benefit liabilities under the Plan, and had purchased annuities to provide benefits; the Employer and the Predecessor Employer shall be treated as if they were a single Employer immediately prior to such event and as unrelated Employers immediately after the event; and if the event giving rise to the predecessor relationship is a benefit transfer, the transferred benefits shall be excluded in determining the benefits provided under the Plan of the Predecessor Employer.
- (5) <u>Special Rules</u>. The limitations of this Section shall be determined and applied taking into account the rules in Regulation Section 1.415(f)-1(d), (e) and (h).
- (6) Aggregation with Multiemployer Plans. If the Employer maintains a multiemployer plan, as defined in Code Section 414(f) and the multiemployer plan so provides, only the benefits under the multiemployer plan that are provided by the Employer shall be treated as benefits provided under a plan maintained by the Employer for purposes of this Section. If accruals under this Plan would otherwise exceed the foregoing limits, then the rate of benefit accrual under this Plan shall be reduced to the extent necessary so that the total Annual Benefits payable at any time under all relevant defined benefit plans will not exceed the Maximum Permissible Benefit.

5.10 <u>Distribution Restrictions</u>.

- (a) <u>Distribution Requirements for Calendar Years Prior to 2002</u>. F-3 and F-3A of Section 1.401(a)(9)-1 of the 1987 proposed Treasury regulations shall apply for purposes of determining minimum required distributions for calendar years prior to 2002.
- (b) <u>Distribution Requirements for Calendar Years 2002 through 2005</u>. The minimum distribution requirements of Code Section 401(a)(9) that were proposed in January 2001 shall apply for purposes of determining minimum required distributions for calendar years 2002, 2003, 2004 and 2005.
- (c) <u>Distribution Requirements for Calendar Years Beginning After December 31, 2005</u>. This Section 5.9(c) shall apply for calendar years beginning after December 31, 2005.
- (1) <u>Precedence and Treasury Regulations Incorporated</u>. Subject to the joint and survivor annuity requirements that are set forth in this Article V, the requirements of subsection 5.9(c) shall take precedence over any inconsistent provisions of the Plan. All distributions required under this Section will be determined and made in accordance with the Treasury regulations under Code Section 401(a)(9), including the incidental death benefit requirement of Code Section 401(a)(9)(G) and the regulations thereunder.
- (2) <u>Limitation on Distribution Periods</u>. As of the first distribution calendar year, distributions, if not made in a single sum, may only be made over one of the following periods (or a combination thereof):

- [a] the life of the Participant;
- [b] the life of the Participant and a designated beneficiary;
- [c] a period certain not extending beyond the Life Expectancy

of the Participant; or

- [d] a period certain not extending beyond the joint and last survivor expectancy of the Participant and the designated beneficiary.
- (3) <u>TEFRA Section 242(b)(2) Elections</u>. Notwithstanding the other provisions of this Section, other than subsection (c)(2), distributions may be made under a designation made before January 1, 1984, in accordance with Section 242(b)(2) of the Tax Equity and Fiscal Responsibility Act (TEFRA) and the provisions of the Plan that relate to Section 242(b)(2) of TEFRA.

(4) Time and Manner of Distribution.

- [a] <u>Required Beginning Date</u>. The Participant's entire interest will be distributed, or begin to be distributed, to the Participant no later than the Participant's required beginning date.
- [b] <u>Death of Participant Before Distributions Begin</u>. If the Participant dies before distributions begin, the Participant's entire interest will be distributed, or begin to be distributed, no later than as follows:
- [i] if the Participant's surviving Spouse is the Participant's sole designated beneficiary, then, except as provided in the adoption agreement, distributions to the surviving Spouse will begin by December 31 of the calendar year immediately following the calendar year in which the Participant died, or by December 31 of the calendar year in which the Participant would have attained age 72, if later.
- [ii] if the Participant's surviving Spouse is not the Participant's sole designated beneficiary, then, except as provided in Section 5.10(c)(9), distributions to the designated beneficiary will begin by December 31 of the calendar year immediately following the calendar year in which the Participant died.
- [iii] if there is no designated beneficiary as of December 31 of the year following the year of the Participant's death, the Participant's entire interest will be distributed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
- [iv] if the Participant's surviving Spouse is the Participant's sole designated beneficiary and the surviving Spouse dies after the Participant but before distributions to the surviving Spouse begin, this subsection [b] other than subsection [b][ii], will apply as if the surviving Spouse were the Participant.

For purposes of this subsection [b] and subsection (c)(7), distributions are considered to begin on the Participant's required beginning date (or, if [iv] above

applies), the date distributions are required to begin to the surviving Spouse under subsection [i] above. If annuity payments irrevocably commence to the Participant before the Participant's required beginning date (or to the Participant' surviving Spouse before the date distributions are required to begin to the surviving Spouse under subsection [i] above, the date distributions are considered to begin is the date distributions actually commence.

[c] Form of Distribution. Unless the Participant's interest is distributed in the form of an annuity purchased from an insurance company or in a single sum on or before the required beginning date, as of the first distribution calendar year distributions will be made in accordance with Sections 5.10(c)(5), (6) and (7). If the Participant's interest is distributed in the form of an annuity purchased from an insurance company, distributions thereunder will be made in accordance with the requirements of Code Section 401(a)(9) and the Treasury regulations. Any part of the Participant's interest which is in the form of an individual account described in Code Section 414(k) will be distributed in a manner satisfying the requirements of Code Section 401(a)(9) and the Treasury regulations that apply to individual accounts.

(5) Determination of Amount to be Distributed Each Year.

[a] <u>General Annuity Requirements</u>. If the Participant's interest is paid in the form of annuity distributions under the Plan, payments under the annuity will satisfy the following requirements:

[i] the annuity distributions will be paid in periodic payments made at intervals not longer than one year;

[ii] the distribution period will be over a life (or lives) or over a period certain not longer than the period described in Section 5.10(c)(6) or (7);

[iii] once payments have begun over a period certain, the period certain will not be changed even if the period certain is shorter than the maximum permitted;

[iv] payments will either be nonincreasing or increase only as follows:

a. by an annual percentage increase that does not exceed the annual percentage increase in a cost of living index that is based on prices of all items and issued by the Bureau of Labor Statistics;

b. To the extent of the reduction in the amount of the Participant's payments to provide for a survivor benefit upon death, but only if the beneficiary whose life was being used to determine the distribution period described in subsection (c)(6) dies or is no longer the Participant's beneficiary pursuant to a qualified domestic relations order within the meaning of Code Section 414(p);

c. to provide cash refunds of employee contributions upon the Participant's death; or

d. to pay increased benefits that result from a

Plan amendment.

[b] Amount Required to be Distributed by Required Beginning Date. The amount that must be distributed on or before the Participant's required beginning date (or, if the Participant dies before distributions begin, the date distributions are required to begin under subsection (4)[b][i] or [ii] is the payment that is required for one payment interval. The second payment need not be made until the end of the next payment interval even if that payment interval ends in the next calendar year. Payment intervals are the periods for which payments are received (e.g., bi-monthly, monthly, semi-annually, or annually). All of the Participant's benefit accruals as of the last day of the first distribution calendar year will be included in the calculation of the amount of the annuity payments for payment intervals ending on or after the Participant's required beginning date.

[c] <u>Additional Accruals After First Distribution Calendar Year</u>. Any additional benefits accruing to the Participant in a calendar year after the first distribution calendar year will be distributed beginning with the first payment interval ending in the calendar year immediately following the calendar year in which such amount accrues.

(6) <u>Requirements For Annuity Distributions That Commence During Participant's Lifetime.</u>

[a] <u>Joint Life Annuities Where the Beneficiary Is Not the Participant's Spouse</u>. If the Participant's interest is being distributed in the form of a joint and survivor annuity for the joint lives of the Participant and a non-spouse beneficiary, annuity payments to be made on or after the Participant's required beginning date to the designated beneficiary after the Participant's death must not at any time exceed the applicable percentage of the annuity payment for such period that would have been payable to the Participant using the table set forth in Q&A 2 of Section 1.401(a)(9)-6 of the Treasury regulations. If the form of distribution combines a joint and survivor annuity for the joint lives of Participant and a non-spouse beneficiary and a period certain annuity, the requirement in the preceding sentence will apply to annuity payments to be made to the designated beneficiary after the expiration of the period certain.

[b] Period Certain Annuities. Unless the Participant's Spouse is the sole designated beneficiary and the form of distribution is a period certain and no life annuity, the period certain for an annuity distribution commencing during the Participant's lifetime may not exceed the applicable distribution period for the Participant under the Uniform Lifetime Table set forth in Section 1.401(a)(9)-9 of the Treasury regulations for the calendar year that contains the annuity starting date. If the annuity starting date precedes the year in which the Participant reaches age 70, the applicable distribution period for the Participant is the distribution period for age 70 under the Uniform Lifetime Table set forth in Section 1.401(a)(9)-9 of the Treasury regulations plus the excess of 70 over the age of the Participant as of the Participant's birthday in the year that contains the annuity starting date. If the Participant's Spouse is the Participant's sole designated beneficiary and the form of distribution is a period certain and no life annuity, the period certain may not exceed the longer of the Participant's applicable distribution period, as determined under this subsection (6)[b], or the joint life and last survivor expectancy of the Participant and the Participant's Spouse as determined under the Joint and Last Survivor Table set forth in Section 1.401(a)(9)-9 of the Treasury regulations, using the Participant's and Spouse's attained ages as of the Participant's and Spouse's birthdays in the calendar year that contains the annuity starting date.

- (7) <u>Requirements For Minimum Distributions Where Participant Dies</u> <u>Before Date Distributions Begin.</u>
- [a] Participant Survived by Designated Beneficiary. Except as provided in Section 5.10(c)(9), if the Participant dies before the date distribution of his or her interest begins and there is a designated beneficiary, the Participant's entire interest will be distributed, beginning no later than the time described in subsection (4)[a] or [b], over the life of the designated beneficiary or over a period certain not exceeding:
- [i] unless the annuity starting date is before the first distribution calendar year, the life expectancy of the designated determined using the beneficiary's age as of the beneficiary's birthday in the calendar year immediately following the calendar year of the Participant's death; or
- [ii] if the annuity starting date is before the first distribution calendar year, the life expectancy of the designated beneficiary determined using the beneficiary's age as the beneficiary's birthday in the calendar year that contains the annuity starting date.
- [b] <u>No Designated Beneficiary</u>. If the Participant dies before the date distributions begin and there is no designated beneficiary as of September 30 of the year following the year of the Participant's death, distribution of the Participant's entire interest will be completed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
- [c] <u>Death of Surviving Spouse Before Distributions to Surviving Spouse Begin</u>. If the Participant dies before the date distribution of his or her interest begins, the Participant's surviving Spouse is the Participant's sole designated beneficiary, and the surviving Spouse dies before distributions to the surviving Spouse begin, this Section 5.10 will apply as if the surviving Spouse were the Participant, except that the time by which distributions must begin will be determined without regard to subsection (4)[a].

(8) Definitions.

- [a] <u>Designated Beneficiary</u>. The individual who is designated as the beneficiary is the designated beneficiary under Code Section 401(a)(9) and Section 1.401(a)(9)-4, Q&A-1, of the Treasury regulations.
- [b] <u>Distribution Calendar Year</u>. A calendar year for which minimum distribution is required. For distributions beginning before the Participant's death, the first distribution calendar year is the calendar year immediately preceding the calendar year which contains the Participant's required beginning date. For distributions beginning after the Participant's death, the first distribution calendar year is the calendar year in which distributions are required to begin pursuant to Section 5.10(c)(4).
- [c] <u>Life Expectancy</u>. Life expectancy as computed by use of the Single Life Table in Section 1.401(a)(9)-9 of the Treasury regulations.

[d] <u>Required Beginning Date</u>. The Required Beginning Date of a Participant is April 1 of the calendar year following the later of the calendar year in which the Participant attains age 72 or the calendar year in which the Participant retires.

(9) <u>Special Elections</u>.

- Beneficiaries. If the Participant dies before distributions begin and there is a designated beneficiary, distribution to the designated beneficiary is not required to begin by the date specified in subsection (c)(4), but the Participant's entire interest will be distributed to the designated beneficiary by December 31 of the calendar year containing the fifth anniversary of the Participant's death. If the Participant's surviving Spouse is the Participant's sole designated beneficiary and the surviving Spouse dies after the Participant but before distributions to either the Participant or the surviving Spouse begin, this election will apply as if the surviving Spouse were the Participant.
- Year Rule. Participants or beneficiaries may elect on an individual basis whether the 5 year rule or the life expectancy rule in Sections 5.10(c)(4) and (7) applies to distributions after the death of a Participant who has a designated beneficiary. The election must be made no later than the earlier of September 30 of the calendar year in which distribution would be required to begin under Section 5.10(c)(4), or by September 30 of the calendar year which contains the fifth anniversary of the Participant's (or, if applicable, surviving Spouse's) death. If neither the Participant nor beneficiary makes an election under this paragraph, distributions will be made in accordance with Sections 5.10(c)(4) and (7) and, if applicable, the election in subsection (9)[a] above.
- 5.11 <u>Cost of Living Adjustment</u>. The Normal Retirement, Early Retirement, Disability Retirement or death benefit payable to an eligible Participant or to the Beneficiary of a deceased eligible Participant shall be increased annually by a post-retirement cost of living adjustment (the "COLA"), calculated as set forth below; provided, however, that the annual increase under this Section shall not apply to a PartTime Emp Participant.
- (a) <u>Teamsters Participants</u>, <u>General Electric Participants</u>, <u>Non-Union Participants</u>, and <u>Dispatchers Participants</u>. With regard to Teamsters Participants, General Electric Participants, Non-Union Participants, and Dispatchers Participants, the additional retirement allowance representing the COLA amount shall be an amount determined by multiplying (i) one percent (1%), by (ii) the amount of the monthly benefit payment determined at the time of the Participant's retirement (including all adjustments for early retirement and form of payment), with no prior adjustments under the provisions of COLA included.
- (b) <u>Water/Wastewater Teamsters Participants</u>. With regard to Water/Wastewater Teamsters Participants, the additional retirement allowance representing the COLA amount shall be an amount determined by multiplying (i) two and a half percent (2.5%), by (ii) the amount of the monthly benefit payment determined at the time of the Participant's retirement (including all adjustments for early retirement and form of payment), with no prior adjustments under the provisions of COLA included.
- 5.12 <u>Employment Termination Prior to Effective Date</u>. Notwithstanding anything in this Plan document to the contrary, the form and amount of the monthly retirement benefit of each

Participant who terminated employment with the Employer prior to the Effective Date shall be determined in accordance with the Prior Plan document that was in effect at the time of the Participant's employment termination.

ARTICLE VI - PAYMENTS FROM THE TRUST FUND

- 6.1 <u>Pension Benefit Payments</u>. Pension benefit payments shall be made to an eligible Participant or his Beneficiary from the Fund in such form or forms as the Participant has elected commencing on the first day of the month following the Participant's date of retirement. Such benefits may be provided through the purchase of an annuity policy from an insurance company in the Trustee's sole discretion.
- 6.2 <u>Death Benefit Payment</u>. If a Participant dies before benefits under the Plan have commenced, death benefit payments (if any) shall be made to the Participant's eligible beneficiary in accordance with Section 5.5. If a Participant dies after distribution of his or her interest has commenced, the remaining portion of such interest will continue to be distributed at least as rapidly as under the method of distribution being used prior to the Participant's death.
- 6.3 <u>Reemployed Retired Participant</u>. Benefit payments to any individual who is receiving a benefit under the Plan and is subsequently rehired by the Employer shall continue during that individual's period of reemployment with the Employer. The monthly benefit that is payable upon a Participant's subsequent employment termination date shall be reduced by the Actuarial Equivalent of any benefit payments (other than disability retirement benefit payments) that such Participant received prior to his or her Normal Retirement Date.
- 6.4 <u>Non-Duplication of Benefits</u>. Notwithstanding any provision of the Plan to the contrary, if any Participant becomes a former Employee due to a severance from employment with the Employer and again becomes a Participant, such renewed participation shall not result in duplication of benefits. Accordingly, if such Participant has received a distribution of all or a portion of his or her Accrued Benefit, then the normal retirement benefit and Accrued Benefit shall be actuarially reduced by the amount of such distribution.
- 6.5 <u>Benefits Payable to Incompetents</u>. If the Plan Administrator shall find that any person to whom a benefit is payable is unable to care for his or her affairs because of accident or illness, or is a minor, the Plan Administrator shall pay the benefit pursuant to whatever is ordered by a court of competent jurisdiction.

6.6 Nonalienation of Benefits.

- (a) <u>In General</u>. The Fund shall not in any manner be liable for or subject to the debts or liability of any Participant or retired Participant. No right or benefit payable under the Plan shall ever be subject in any manner to alienation, sale, transfer, assignment, pledge or encumbrance of any kind either voluntarily or involuntarily.
- (b) <u>Exception for EDROs</u>. Subsection (a) shall not apply to an "eligible domestic relations order" defined in Act 46 of 1991, Public Employees Retirement System Eligible Domestic Relations Order Act of Michigan. The administrator shall establish a written procedure to determine the eligible status of domestic relations orders and to administer distributions under such eligible orders. Further, to the extent provided under an "eligible domestic relations order,"

a former spouse of a Participant shall be treated as a spouse or surviving spouse for all purposes under the Plan.

(c) Exception for Certain Debts to Plan. Subsection (a) shall not apply to an offset to a Participant's Accrued Benefit against an amount that the Participant is ordered or required to pay to the Plan with respect to a judgment, order, or decree issued, or a settlement entered into in accordance with Code Sections 401(a)(13)(C) and (D).

ARTICLE VII - FINANCING

7.1 <u>Trust Fund</u>. All contributions made under this Plan shall be paid to the Trustee and deposited in the Fund. Except as otherwise provided in Section 9.6 below, all assets of the Fund, including investment income, shall be retained for the exclusive benefit of Participants and their beneficiaries, shall be used to pay benefits to such persons or to pay administrative expenses to the extent not paid by the Employers, shall not be diverted to or used for purposes other than for the exclusive benefit of the Participants in the Plan or their beneficiaries, and shall not revert to or inure to the benefit of any Employer. Notwithstanding anything herein to the contrary, on the request of the Employer, a contribution which was made by a mistake of fact shall be returned to the requesting entity within one year after the payment of the contribution, or the disallowance of the deduction (to the extent disallowed).

The Named Fiduciary shall exercise the power to manage and control the assets of the Trust which are held for the benefit of Plan Participants.

7.2 <u>Employer Contributions</u>. The Employer shall make contributions in such amounts and at such times as the Employer shall determine to be necessary to provide the benefits under the Plan determined by the application accepted actuarial methods and assumptions. The method of funding shall be consistent with Plan objectives.

Forfeitures arising under the Plan because of severance of employment before a Participant becomes eligible for a pension hereunder or for any other reason shall be applied to reduce the cost of the Plan, and not to increase the benefits otherwise payable to Participants.

7.3 <u>Plan-to-Plan Transfers from Qualified Plans</u>. With the consent of the Sponsor, amounts may be transferred to this Plan from other tax qualified plans under Code Section 401(a), provided the plan from which such funds are transferred permits the transfer to be made and the transfer will not jeopardize the tax-exempt status of the Plan or Trust or create adverse tax consequences for the Employer.

ARTICLE VIII - ADMINISTRATION OF THE PLAN

8.1 <u>Plan Administrator and Designation of Administrative Authority</u>. The Plan Administrator shall be the Sponsor. The Sponsor may appoint any person, including, but not limited to, the Employees of the Employer, to perform the duties of the Plan Administrator. Any person so appointed shall signify acceptance by filing written acceptance with the Sponsor. Upon the resignation or removal of any individual performing the duties of the Plan Administrator, the Sponsor may designate a successor.

8.2 Responsibilities. The Employer, Plan Administrator, Trustee and Investment Advisor shall have only those specific powers, duties, responsibilities and obligations as are specifically given them under this Plan or any other contract related hereto. The Employer shall have the sole responsibility for making the contributions that are necessary to provide benefits under the Plan, and shall have the sole authority to appoint and remove the Trustee, Plan Administrator and any Investment Advisor that may be provided for under the Trust, and to amend or terminate, in whole or in part, this Plan or the Trust. The Plan Administrator shall have the sole responsibility for the administration of this Plan, which responsibility is specifically described in this Plan, the Trust and administrative procedures, if any. The Trustee shall have the sole responsibility for the administration of the Trust and the management of the assets held under the Trust, all as specifically provided in the Trust and/or other contract.

The Employer, Plan Administrator, Trustee and Investment Advisor each warrants that any directions given, information furnished, or action taken by it shall be in accordance with the provisions of the Plan and/or other related contract, as the case may be, that authorizes or provides for such direction, information or action. Furthermore, each may rely upon such direction, information or action as being proper under this Plan, and is intended under this Plan that each shall be responsible for the proper exercise of its own responsibilities and obligations under this Plan and shall not be responsible for any act or failure to act of another. No fiduciary guarantees the Fund in any manner against investment loss or depreciation in asset value.

- 8.3 <u>Records and Reports</u>. The Plan Administrator shall keep a record of all actions taken and shall keep all other books of account, records, policies, and other data that may be necessary for proper administration of the Plan and shall be responsible for supplying all information and reports to the Internal Revenue Service, Participants, Beneficiaries and others as required by law.
- 8.4 Other Plan Administrator Powers and Duties. The Plan Administrator shall have such duties and powers as may be necessary to discharge its duties hereunder, including but not by way of limitation, the duties and powers described below.
- (a) The Plan Administrator shall have authority to construe and interpret the Plan, decide all questions of eligibility and determine the amount, manner and time of payment of any benefits hereunder.
- (b) The Plan Administrator shall have authority to prescribe procedures and forms to be followed by Participants or beneficiaries filing applications for benefits.
- (c) The Plan Administrator shall have authority to prepare and distribute, in such manner as the Plan Administrator determines to be appropriate, information explaining the Plan, and to receive from the Participants such information as is necessary for the proper administration of the Plan.
- (d) The Plan Administrator shall have authority to appoint or employ individuals to assist in the administration of the Plan and any other agents it deems advisable, including one or more investment managers (pursuant to the terms of the Trust) and legal and actuarial counsel and to pay the cost associated with such appointment or employment from Plan assets.

(e) The Plan Administrator shall have authority to receive, evaluate and act on all claims for benefits under the Plan in accordance with the procedure described in this subsection (e).

The Plan Administrator shall have discretionary authority and shall make all determinations as to the right of any person to a benefit. The Plan Administrator shall make each claim determination in a uniform and nondiscriminatory manner, provided that such determination shall be made within 90 days after the receipt of the claim by the Plan Administrator. The Plan Administrator shall either grant the claim, deny the claim, or notify the Participant, former Participant, or Beneficiary (hereafter "Claimant") that special circumstances have required the extension of time for the processing of the claim, such extension not to exceed 180 days from the original notice.

Within 30 days after denial of any benefit under the Plan, the Plan Administrator shall give to the Claimant written notice by certified mail, directed to their last address of record with the Plan Administrator, of the denial of claim for benefits. The notice shall set forth the specific reasons for such denial, shall make specific reference to Plan provisions upon which the denial is based, shall describe any additional material or information necessary for the Claimant to perfect their claim and why such material or information is necessary, and shall advise the Claimant that they may file a written appeal of the determination with the Plan Administrator within one hundred eighty (180) days after receipt of such Notice. In connection with such appeal, the Claimant or their duly authorized representative may review pertinent documents and submit issues and comments in writing. Failure of the Claimant to file written appeal with the Plan Administrator within the allowable 180-day period shall constitute an irrevocable consent by the Claimant to the Plan Administrator's decision denying the benefit claimed, and the Plan Administrator's written notice shall so state. Within 60 days after receipt of the request for review and receipt of all documents needed for the review, the Plan Administrator shall notify the Claimant either as to the decision on the appeal or that special circumstances require an extension of time for processing, such extension not to exceed 120 days from the date of the filing of the appeal. The claim appeal shall be reviewed by a fiduciary who is neither the party who made the initial adverse determination nor a subordinate of that party. The review may not afford deference to the initial denial of benefits and must take into account all comments, documents, records and other information submitted by the claimant, without regard to whether such information was previously submitted or relied upon in the initial determination.

- (f) The Plan Administrator shall have authority to arrange for all services necessary for the proper administration of the Plan and to pay, from Plan assets, reasonable compensation pursuant to a reasonable contract for all such services. The Plan Administrator may arrange for any or all of such services to be supplied by the Employer and pay such compensation from Plan assets to the Employer in accordance with the requirements of relevant law.
- 8.5 <u>Rules and Decisions</u>. The Plan Administrator may adopt such rules as it deems necessary, desirable and appropriate. All rules and decisions of the Plan Administrator shall be uniformly and consistently applied to all Participants in similar circumstances. When making a determination or calculation, the Plan Administrator shall be entitled to rely upon information that is furnished by a Participant or beneficiary, the legal counsel to the Plan, the Actuary, the Plan accountants, the Trustee, the Investment Advisors or other consultants.

- 8.6 <u>Authorization of Benefit Payments</u>. The Plan Administrator shall issue directions to the directed Trustee concerning all benefits that are to be paid from the Fund pursuant to the provisions of the Plan and warrants that all such directions are in accordance with this Plan.
- 8.7 <u>Application and Forms for Pension</u>. The Plan Administrator may require a Participant or beneficiary to complete and file with the Plan Administrator an application for benefits and all other forms that are approved by the Plan Administrator and to furnish all pertinent information, including proof of age, that is requested by the Plan Administrator. The Plan Administrator may rely upon all such information so furnished, including the Participant's or beneficiary's current mailing address.
- 8.8 Facility of Payment. Whenever, in the opinion of the Plan Administrator, a person who is entitled to receive any payment of a benefit or installment thereof from this Plan is under a legal disability or is incapacitated in any way so as to be unable to manage his financial affairs, the Trustee may be given instructions to make payments to such person or to his or her legal representative or to a relative or friend of such person for his or her benefit, or the Plan Administrator may instruct the directed Trustee to apply the payment for the benefit of such person in such manner as the Authorized Employer Representative considers advisable. Any payment of a benefit or installment thereof in accordance with the provisions of this Section shall be a complete discharge of any liability for the making of such payment under the provisions of the Plan.
- 8.9 <u>Indemnification</u>. The governing body of the Employer, the Plan Administrator, and the individual members thereof shall be indemnified by the Employer and not from the Fund against any and all liabilities arising by reason of any act or failure to act that is made in good faith pursuant to the provisions of the plan, including expenses that are reasonably incurred in the defense of any claim relating thereto.

ARTICLE IX - AMENDMENT AND TERMINATION OF PLAN

- 9.1 <u>Right to Terminate</u>. In accordance with the procedures set forth in this Article IX and applicable law, the Sponsor may terminate the Plan in whole or in part at any time. In the event of the dissolution, merger, consolidation or reorganization of the Sponsor, the Plan shall terminate and the Fund shall be dealt with as set forth in this Article IX unless the Plan is continued by a successor to the Sponsor, pursuant to Section 9.8.
- 9.2 <u>Right to Amend</u>. The Sponsor reserves the right to make from time to time any amendment or amendments to this Plan which do not cause any part of the Fund to be used for, or diverted to, any purpose other than the exclusive benefit of Participants or their beneficiaries.
- 9.3 <u>Effect of Amendment or Termination</u>. No amendment of the Plan shall deprive any person who has a vested interest in the Plan of any such vested interest unless such deprivation shall be required to qualify or maintain qualification of the Plan under the Internal Revenue Code. No amendment or termination of the Plan shall cause any part of the Fund to revert to the Sponsor except as provided under Section 9.6 below. Upon termination or partial termination of the Plan, each affected Participant shall become fully vested in his or her Accrued Benefit. In the event of merger or consolidation with another plan or transfer of assets or liabilities to another plan, each Participant shall be entitled to the same or greater benefit under such other plan as he or she would have been entitled to under this Plan if this Plan had terminated at such time. This Section 9.3 shall be administered in accordance with the Code and applicable Michigan law.

- 9.4 <u>Allocation of the Fund on Termination or Partial Termination</u>. In the event of termination or partial termination of the Plan, the assets then remaining in the Fund, after providing the expenses of the Plan, shall be allocated, to the extent they shall be sufficient, for the purpose of paying benefits in the order of precedence and in the manner set forth in Section 9.5 hereof.
- Disposition of Assets on Termination or Partial Termination. Upon any termination (full or partial), all amounts shall be allocated in accordance with the provisions hereof and the Accrued Benefit, to the extent funded as of such date, of each affected Participant shall become fully vested and shall not thereafter be subject to forfeiture. However, Participants who were not fully vested at the time they received a complete distribution of their vested benefits prior to the date of termination, shall not become entitled to any additional vested benefits on account of Plan termination. The preceding sentence does not apply to Participants who are affected by a partial termination by operation of law. Upon full termination of the Plan, the Employer shall direct the distribution of the assets in the Trust Fund to the Participants in a manner which is consistent with the terms of this Plan, the Code, and relevant Michigan law. In such case, the Trustee shall distribute the assets to the remaining Participants in the Plan and to retired Participants in cash or through the purchase of irrevocable deferred commitments from an insurer, subject to provision for expenses of administration or liquidation. Such distributions shall be allocated in the following order to the extent of the sufficiency of such assets, basing such allocation on the Accrued Benefit for each such Participant at the date of termination of the Plan: (1) to provide pensions to retired Participants who have retired under the Plan prior to its termination without reference to the order of retirement; (2) to provide normal retirement benefits to Participants who have reached their Normal Retirement Dates but have not retired on the date of termination, without reference to the order in which they shall have reached their Normal Retirement Date; and (3) to provide normal retirement benefits to Participants who have not yet reached their Normal Retirement Date on the date of termination, without reference to the order in which they will reach their Normal Retirement Date. Such benefits will be based upon Accrued Benefits as of the date of termination.

If the assets of the Fund that are applicable to any of the foregoing categories are insufficient to provide full benefits for all persons in such group, the benefits that are otherwise payable to such individuals shall be reduced proportionately. The Actuary shall calculate the allocation of the assets of the Fund in accordance with the above priority categories and certify his calculations to the Plan Administrator.

- 9.6 <u>Residual Amounts</u>. In no event shall the Employer receive any amounts from the Trust Fund upon termination of the Plan, except that, and notwithstanding any other provision of the Plan, it shall receive such amounts, if any, as may remain after the satisfaction of all liabilities of the Plan and arising out of any variations between actual requirements and expected actuarial requirements.
- 9.7 <u>No Additional Liability</u>. Upon termination of the Plan, a Participant shall have recourse toward satisfaction of his or her nonforfeitable benefit only from the Plan assets, subject to applicable Michigan law.
- 9.8 <u>Successor Sponsor</u>. In the event of the dissolution, merger, consolidation or reorganization of the Employer, provision may be made by which the Plan and Trust will be continued by its successor; and, in that event, such successor shall be substituted for the Employer under the Plan. The substitution of the successor shall constitute an assumption of the Employer's Plan liabilities

by the successor and the successor shall have all the powers, duties, and responsibilities of the Employer under the Plan.

- 9.9 <u>Plan Assets</u>. In the event of any merger or consolidation of the Plan, or transfer in whole or in part of the assets and liabilities of the Trust Fund to another trust fund held under any other plan of deferred compensation maintained or to be established for the benefit of all or some of the Participants of this Plan, the assets of the Trust Fund applicable to such Participants shall be transferred to the other trust fund only if:
- (a) each Participant would be entitled to a benefit immediately after the merger, consolidation or transfer which is equal to or greater than the benefit he or she would have been entitled to immediately before the merger, consolidation or transfer (if this Plan had then terminated);
- (b) the Sponsor shall comply with reporting and disclosure requirements applicable to the proposed transaction;
- (c) resolutions of the governing body of the relevant Employer under this Plan, and any new or successor employer of the affected Participants, shall authorize such transfer of assets; and, in the case of the new or successor employer of the affected Participants, its resolutions shall include an assumption of liabilities with respect to such Participants' inclusion in the new employer's plan, and
- (d) such other plan and trust are qualified under Code Sections 401(a) and 501(a).

ARTICLE X - MISCELLANEOUS

- 10.1 <u>Unclaimed Pension Checks</u>. If a check in payment of a benefit payable under this Plan has been mailed by regular United States mail to the last address of the payee furnished to the Plan and the check is returned unclaimed, payments to such payee shall be discontinued and shall be held by the Plan while the Plan Administrator conducts a search for the payee. If the Plan Administrator is unable to locate the payee after such a search, then the amount held in the Plan for said payee shall be forfeited. Said forfeiture shall occur as of the last day of the Plan Year in which the Plan Administrator concludes its search without locating the payee. If the payee later files a claim for benefits with the Plan, the Employer must promptly reinstate the forfeited benefit on behalf of the payee. Payment of the restored benefit to said payee will commence as soon as reasonably possible after the Plan Administrator determines that the Plan authorizes the payment.
- 10.2 <u>Correction of Errors</u>. Notwithstanding any other provision of this Plan to the contrary, the Sponsor and the Plan Administrator reserve the right to correct (retroactively, if necessary) any error in the Plan language or in the administration of the Plan which was inadvertently made in the good faith creation and/or administration of this Pension program.
- 10.3 <u>Nonguarantee of Employment</u>. Nothing contained in this Plan shall be construed as a contract of employment between the Employer and any Participant, or as a right of Participant to be continued in the employment of the Employer, or as a limitation of the right of the Employer to discharge any of its Participants, with or without cause.

10.4 <u>Gender and Number</u>. Whenever applicable, the masculine pronoun shall include the feminine pronoun, and the singular shall include the plural.

CITY OF ESCANABA

Dated:	By:	
	•	

City of Escanaba Defined Benefit Retirement Plan Board RULES OF PROCEDURE

1. AUTHORITY/PURPOSE

- 1.1. **General.** The Escanaba City Council created the City of Escanaba Defined Benefit Retirement Plan Board. The City of Escanaba Defined Benefit Retirement Plan Board is charged with oversight of the pension plan, established under the authority of Public Employee Retirement System Act (PERISA), and Michigan Public Act 314 of 1965 as amended.
- 1.2. **Membership.** The plan is governed by a five-member board consisting of the City Manager, City Controller, City Treasurer, one member of the City Council, and one member of the public with preference to any plan member.
- 1.3. The non-administrative members of the Board shall be appointed by the City Council with general concurrence.
- 1.4. **Statement of Investment Policy.** The Board shall adhere to the Statement of Investment Policy, as approved by the City Council. Any amendment to the Statement of Investment Policy shall be referred to the City Council for approval.

2. GENERAL RULES

- 2.1 **Meetings to be Public.** All meetings are subject to the Michigan Open Meetings Act.
- 2.2 **Quorum.** Three (3) members shall constitute a quorum necessary for the transaction of business. If a quorum is not present, those in attendance will be named and they shall adjourn to a later time.
- 2.3 **Journal of Proceedings.** An account of all proceedings shall be kept and shall be entered in a book constituting the official record and deposited to the City Clerk.
- 2.4 **Right of Floor.** Any member desiring to speak shall be recognized by the Chair and confine remarks to the subject under consideration.
- 2.5 **Election of Officers.** At the first regularly scheduled meeting of each year, the City of Escanaba Defined Benefit Retirement Plan Board shall hold an election of Officers for the positions of Chair and Vice Chair.

3. MEETINGS

- 3.1. **Public Notice of Meetings.** The City Clerk shall be responsible for providing the proper notice of all meetings as specified by the State Open Meetings Act.
- 3.2. **Regular Meetings.** The City of Escanaba Defined Benefit Plan Board shall meet semi-annually to review the annual actuarial valuation and review investment performance, and to attend to other business, upon the call of City Administration in collaboration with the Chair.

- 3.3. **Special Meetings.** Special meetings may be called by the City Administration in collaboration with the Chair. The call for a special meeting shall specify the day, the hour and the location. The City Clerk shall give public notice as specified by the State Open Meetings Act.
- 3.4. **Cancelled Meetings.** Any meeting may be cancelled to a later date and time provided that no adjournment shall be for a longer period than until the next scheduled meeting.

4. CHAIR AND DUTIES

- 4.1 **Chair.** The Chair, if present, shall preside as Chair at all meetings. In the absence of the Chair, the Vice Chair shall preside. In the absence of both the Chair and the Vice Chair, the Board shall elect a temporary Chair.
- 4.2 **Call to Order.** The meetings shall be called to order by the Chair or, in their absence, by the Vice Chair. In the absence of both the Chair and the Vice Chair, the meeting shall be called to order by the City Treasurer for the election of a temporary Chair.
- 4.3 **Preservation of Order.** The Chair shall preserve order and decorum; prevent attacks on personalities or the impugning of members' motives, and confine members in debate to the question under discussion.

5. ORDER OF BUSINESS AND AGENDA

- 5.1 **Order of Business.** The general rule as to the order of business in regular meetings shall be as follows:
- 1. Call to Order
- Roll Call
- 3. Approval and/or Correction to Minutes of Previous Meeting
- 4. Approval/Adjustments to the Agenda
- Conflict of Interest Declarations
- 6. Public Comment
- 7. Unfinished Business
- 8. New Business
- 9. General Public Comment
- 10. Board/Administration Comments/Announcements
- 11. Adjournment
- 5.2 **Agenda.** The order of business of each meeting shall be contained in the Agenda prepared by the City Treasurer in collaboration with the Chair. Agenda items must be submitted to the City Treasurer ten (10) days prior to the meeting. The Agenda shall be a listing by topics of subjects to be considered by the Board and shall be delivered to members of the Board at least five (5) days prior to the meeting. Draft meeting minutes shall be available for public review within ten (10) business days after the meeting.

- 5.3 **Voting.** A majority vote of those present shall be required to approve a motion, recommendation, or finding.
- 5.4 **Roll Call Votes.** A roll call vote shall be taken at the request of any member or when the Chair cannot determine the results of a voice vote.
- 5.5 **Minutes.** A majority vote consisting of at least a quorum shall be required to approve minutes.

6. <u>CITIZENS' RIGHTS</u>

- 6.1. **Addressing the Board.** Members of the public desiring to address the Board by oral communications on any agenda item shall first secure the permission of the Chair.
- 6.2. **Manner of Addressing the Board.** Each person addressing the Board shall approach the lectern and must give their name in an audible tone of voice for the record. All remarks shall be addressed to the Board as a body and not to any member thereof. No person, other than members of the Board and the person having the floor shall be permitted to enter into any discussion, either directly or through the members. No questions shall be asked except through the Chair.
- 6.3. **Personal and Slanderous Remarks.** Any person making personal, impertinent or slanderous remarks, or who shall become disorderly, while addressing the Board, will be requested to leave the lectern.
- 6.4. **Chair Will Refer Citizens' Complaints.** The Chair will refer citizens' complaints to City Administration. The City Clerk will review all complaints and report their findings to the City of Escanaba Defined Benefit Retirement Plan Board on or before their next regularly schedule meeting.
- 6.5. **Written Communications.** Interested parties may address the Board by written communication in regard to any matter concerning Board business by addressing their concern to the City Manager.

STATEMENT OF INVESTMENT POLICY

CITY OF ESCANABA

DEFINED BENEFIT RETIREMENT PLAN

- I. PURPOSE OF STATEMENT
- II. GENERAL OPERATION OF PLAN
- III. INVESTMENT APPROACH
- IV. INVESTMENT RESTRICTIONS
- V. PERFORMANCE MONITORING
- VI. COMMUNICATION
- VII. SUMMARY

ADOPTED 10/19/2023

I. PURPOSE

The purpose of the City of Escanaba Defined Benefit Retirement Plan (hereinafter referred as "Plan") Investment Policy Statement (herein after referred to as "Statement") is to establish, document and communicate to interested parties, the goals, philosophies, guidelines, objectives, and risk tolerances relating to investment of Plan assets.

II. GENERAL OPERATION OF THE PLAN

- A. The Plan is established under the authority of Michigan Public Act 314 of 1965 as amended, known as the Public Employee Retirement System Investment Act ("Act 314").
- B. The Plan is governed by a five-member board ("Board") consisting of the City Manager, City Controller, City Treasurer, one member of the City Council, and one member of the public with preference to any Plan member. The Board is responsible for oversight and decision making relating to the general operation of the Plan. The Board's oversight and decision making includes, but is not limited to, selection and replacement of investment managers and actuarial consultant, asset allocation policy, maintenance of the Plan's actuarial integrity and on-going evaluation of the Plan's progress toward its various goals and objectives. The Board shall employ the "Prudent Man" doctrine when making decisions concerning the investment of Plan assets. Plan assets shall be managed with the long term interest and integrity of the Plan in mind.
- C. The Plan is funded through contributions by the City of Escanaba.
- D. The City Treasurer is responsible for keeping records of investment activity within the Plan. Such records are made available for public inspection at the Office of the City Treasurer, 410 Ludington, Escanaba, Michigan.
- E. Annually, the Plan is audited by a public accounting firm and evaluated by an actuarial consultant. Audits and actuarial valuations are also available for public inspection at the office of the City Treasurer.

III. INVESTMENT APPROACH

- A. It is the philosophy of the Plan to pursue management of Plan assets through the use of outside investment managers. For purposes of this document, mutual funds or mutual fund companies shall be deemed to be outside managers. Outside managers may also include, but are not limited to, affiliates of registered and licensed brokerage firms, registered professional money management firms, mutual funds, and money market instruments, banks, trust departments, and independent advisors. The Board may employ any number of outside managers it deems necessary for the effective administration of the Plan. The Board shall not undertake to select individual securities as an on-going means to achieve Plan objectives. For purposes of this document, mutual funds are not considered to be individual securities. Instead, selection of individual securities will be left solely to the discretion of investment managers hired by the Plan.
- B. For purposes of administration, the Board shall separate the overall investment portfolio into an equity component and a fixed income component. Each component will be comprised of cash and securities, the proportions of which shall be left to the discretion of the investment manager(s) controlling each respective component. Maximum equity exposure is hereby set at 65% of Plan assets. It shall be the responsibility of the Board to monitor on an on-going basis, the allocation

- of dollars between fixed income and equity components, and to transfer funds between components as deemed necessary to achieve the objectives of the Plan.
- C. Plan assets will be managed in accordance with Michigan Public Act 314 of 1965 as amended. All external money managers are required to abide by all applicable provisions of Act 314 and by all applicable provisions of this Statement. External managers will be furnished with a copy of Act 314 and with a copy of this Statement and will be required to acknowledge in writing the receipt thereof. If a manager is unable to comply with any of the provisions of Act 314 or any provision of this Statement, he or she must communicate as such in writing prior to investing any assets of the Plan. All external managers are required to comply with applicable laws of the State of Michigan, whether now in force, or hereinafter enacted.
- D. Subject to the requirements set forth herein, all external money managers shall have full authority to select individual securities, and to determine an appropriate mix of cash and investments within their allotment of Plan assets.
- E. The Board shall determine if, and when, outside custodians shall be employed. If employed, outside custodians shall perform the following functions:
 - 1. Notify the Plan sponsor of transactions as they occur.
 - 2. Provide electronic quarterly statements detailing specific investment holdings with summaries of gains/losses, income received, and activity for the period covered.
 - 3. Provide safe keeping of assets.
- F. Outside custodians are not required when mutual funds are used. In the event the Board employs a registered mutual fund as an outside manager, the mutual fund's statements and other reporting shall be an acceptable substitute for the services of a custodian.
- G. In addition to the services of professional money managers and custodians, the Board may contract with outside parties for advisory and related services. Examples of such services may include brokerage, performance monitoring, studies, and general consultations. Decisions to contract for such services shall be made on the basis of perceived cost versus benefit.

IV INVESTMENT RESTRICTIONS

- A. Equity investments shall be limited to exchange listed securities which have a readily available market and for which prices are readily available. Exceptions to this provision will be considered when they are brought to the attention of the Board in writing prior to purchase of such securities. This provision shall not apply to registered mutual funds held by the Plan.
- B. Notwithstanding other provisions of this Statement, the following are examples of permitted investments for use by external managers: Treasury securities, agency securities, mortgage-backed instruments, guaranteed investment contracts, corporate bonds, common stocks, preferred stocks, mutual funds, ETFs, money market mutual funds, commercial paper, repurchase agreements and banker's acceptances. Securities not specifically mentioned are permissible investments if they are not prohibited under applicable law or under the provisions of this Statement.
- C. External money managers are required to notify the Board in writing in the event of any change that will have a significant impact on the operations of the firm. Written notice is required in the event of any of the following:
 - 1. Change in ownership of the firm

- 2. Change in portfolio managers
- 3. Change in investment style
- 4. Investigation of the firm or its management by a regulatory body
- 5. Material litigation
- D. Upon written request from the Board, money managers will be required to furnish the Board references. Clients who have requested in writing that their identity not be disclosed may be excluded from such list. This provision does not apply to externally employed, registered, mutual funds.
- E. External money managers shall be required to disclose the existence of party in interest transactions and the corresponding details thereof. Notwithstanding SEC and other requirements, registered mutual funds are exempt from this provision.
- F. The Board shall ensure that portfolios are diversified with respect to risk, and where applicable, maturity. The Board shall not permit any one security to represent more than 5% of the overall fund, and shall not permit one industry to represent more than 30% of the fund. Mutual funds are not to be classified as individual securities for purposes of this provision. External money managers are not individually responsible for adhering to this provision, since they will not have knowledge of holdings outside of their management.
- G. External managers shall disclose to the Board in writing if any individual security exceeds 10% of the assets that they manage on behalf of the Plan. Additionally, external managers shall disclose to the Board in writing if any individual industry represents more than 50% of the assets that they manage on behalf of the Plan. Externally employed, registered mutual funds are exempt from this provision.
- H. The fixed income portfolio must carry an average rating of A or better by Standard & Poor's and Moody's. Money managers may hold below investment grade rated securities as long as the aggregate value of such securities does not exceed 10% of a manager's allotted funds. Mutual fund managers are exempt from this provision.
- I. The Board shall impose the following limitations on foreign securities. The board shall ensure that foreign equities do not exceed 15% of the overall investment portfolio. Foreign fixed income holdings shall not exceed 10% of the overall investment portfolio.

V. PERFORMANCE MONITORING

- A. The Board shall review year to date performance relative to established benchmarks. Performance review shall take place at semiannual regular meetings. It shall be the responsibility of the investment manager to report performance. If desired, the Board shall have the option to seek independent performance review.
- B. The Board shall use a blended portfolio consisting of 85% Russell 3000 Index and 15% MSCI ACWI ex-US Index as its proxy for the equity market. The Board acknowledges that investment managers will have varying disciplines. As such, the Board also acknowledges that the blended equity proxy may not be a relevant indicator in every case. The Board therefore will consider peer group equity comparisons when appropriate.

- C. The Board shall use the Bloomberg US Aggregate Index as its proxy for bond market returns. Additionally, the Board shall evaluate the credit quality and duration of the Plan's fixed income component relative to its market proxy.
- D. In addition to monitoring the total return of investments, the Board shall review investment volatility relative to established benchmarks. Standard deviation shall be the primary measure of volatility.

VI. COMMUNICATION

A. External money managers shall furnish the Board with an electronic quarterly statement containing a list of holdings, a summary of gains and losses, a portfolio valuation, and a summary of activity. Fixed income managers are required to furnish semiannual calculations of duration, average maturity and credit quality. If registered mutual funds are employed by the Board, they shall be responsible only for adhering to their individual regulatory reporting requirements.

VII. SUMMARY

- A. This Statement sets forth the philosophies, goals, objectives, tolerances and methodologies for the investment of Plan assets. Absent provisions to the contrary, all parties involved in the investing of Plan assets are expected to adhere to the provisions found herein.
- B. Should any provision of this Statement be found to be in conflict with Michigan Law, Michigan Law shall preside over the terms of this Statement.

IT IS HEREBY CERTIFIED THAT WE HAVE RECEIVED A COPY OF THE STATEMENT OF INVESTMENT POLICY OF THE CITY OF ESCANABA DEFINED BENEFIT RETIREMENT PLAN. WE HAVE ALSO BEEN FURNISHED WITH A COPY OF MICHIGAN PUBLIC ACT 314. WE FURTHER CERTIFY THAT WE HAVE READ THE AFOREMENTIONED DOCUMENTS AND AGREE TO ABIDE BY THE TERMS AND CONDITIONS CONTAINED THEREIN.

Name of Firm:
Name of Representative:
Name of Representative.
Title of Representative:
Signature of Representative:
Date: